### EUROPEAN CENTRAL BANK

#### **WORKING PAPER SERIES**



**WORKING PAPER NO. 56** 

STABILIZATION POLICY IN A TWO COUNTRY MODEL AND THE ROLE OF FINANCIAL FRICTIONS

### **BY ESTER FAIA**

# April 2001

### **WORKING PAPER SERIES**



### **WORKING PAPER NO. 56**

STABILIZATION POLICY IN A TWO COUNTRY MODEL AND THE ROLE OF FINANCIAL FRICTIONS\*

#### **BY ESTER FAIA\*\***

## April 2001

 I thank William Baumol, Pierpaolo Benigno, Vincent Brousseau, Thomas Cooley, Carsten Detken, Günter Coenen, Mark Gertler, Philipp Hartmann, Tommaso Monacelli, Frank Smets and Pedro Teles, for many useful discussions. I thank all the seminar participants at the European Central Bank and New York University. All errors are mine.
 \*\* Ester Faia, Department of Economics, New York University, 269 Mercer Street, 10012 New York, NY. E-mail address: ef284@nyu.edu.

#### © European Central Bank, 2001

Address	Kaiserstrasse 29
	D-60311 Frankfurt am Main
	Germany
Postal address	Postfach 16 03 19
	D-60066 Frankfurt am Main
	Germany
Telephone	+49 69 1344 0
Internet	http://www.ecb.int
Fax	+49 69 1344 6000
Telex	411 144 ecb d

#### All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged. The views expressed in this paper are those of the authors and do not necessarily reflect those of the European Central Bank.

# Contents

Ab	bstract		5
I	Intr	Introduction	
2	The	Model	12
	2.1	The Aggregate Demand Block 2.1.1 Workers' Behavior in The Home ountry and	13
	2.2	the Foreign Country The Aggregate Supply Block: Firms' Behavior at Home	13
		and in the Foreign County	16
		2.2.1 The Competitive Sector and the Capital Producers	16
		2.2.2 The Behavior of the Monopolistic Sector	18
	2.3	The Financial Intermediary	19
	2.4	Fiscal Authority	20
	2.5 2.6	The Equilibrium Conditions The Loglinearized version of the model	20 21
3	The	Monetary Policy Rules	23
4	Cali	bration	24
5	Fixe	d Versus Floating Exchange Rates	25
6	Managed Exchange Rates Versus Floating and		
	Versus Fixed Exchange Rates		29
7		tility of Variables under Different Regimes	
	and	Welfare Comparison	30
8	Cor	nclusions	33
Ref	erence	S	34
9	Арр	endix	36
	9.1	Derivation of the Complete Loglinearized Model	36
		9.1.1 Aggregate Demand Equations	36
	9.2	The Calvo Producers	40
	9.3	The Financial Intermediary	41
	9.4	The Steady State of the Economy	43
	9.5	The Welfare Measure	43
Tab	oles and	d figures	45
Eur	opean	Central Bank Working Paper Series	56

#### Abstract

This paper studies the optimal choice of exchange rate regimes between two large currency areas. It provides a positive and normative analysis of alternative monetary policy rules in a model with sticky prices, monopolistic competition, and frictions in the processes of capital accumulation and acquisition of external finance. The stabilization and welfare analysis provides a sound result on the desirability of monetary policy and exchange rate flexibility as business cycle smoothing devices. Given the presence of financial frictions the paper gives a richer explanation of the mechanism behind the stabilization properties of floating exchange rates and explains the difference in sign of the international transmission of shocks compared to the model without capital. In a two country model without capital the pattern of output is mainly determined by the pattern of consumption: any movement in the exchange rate under floating exchange rates causes movements in the price of the international traded bond and in consumption and consequently in output. In the model with capital and financial frictions output mimics the movements in investment: an active monetary authority reacting to exchange rate movements generates perverse movements in the interest rate, destabilizing investment and output. The paper also suggests how monetary policy can improve financial stability, stressing the importance of the interest rate smoothing in tuning movements in investment and output and in reducing the welfare cost of financial frictions mostly under fixed exchange rates.

JEL Classification Numbers: E3, E42, E44, E52, F41.

*Keywords:* financial frictions, monetary regimes, financial stability, stabilization policy.

<sup>\*</sup>I thank William Baumol, Pierpaolo Benigno, Vincent Brousseau, Thomas Cooley, Carsten Detken, Günter Coenen, Mark Gertler, Philipp Hartmann, Tommaso Monacelli, Frank Smets and Pedro Teles, for many useful discussions. I thank all the seminar participants at the European Central Bank and New York University. All errors are mine.

<sup>&</sup>lt;sup>†</sup>Ester Faia, Department of Economics, New York University, 269 Mercer Street, 10012 New York, NY. E-mail address: ef284@nyu.edu.

### 1 Introduction

"For the major countries and regions (the United States, the Euro area, and Japan) where unrestricted capital mobility is the established norm, and where pursuit of a common monetary policy appears unlikely to be consistent with the key goals of macroeconomic stability, floating exchange rates will, and should, continue to prevail". Michael Mussa<sup>1</sup>

The central focus of this paper is the optimal choice of foreign exchange system in terms of macroeconomic stability and the international transmission mechanism of shocks between the Euro area and the U.S. economy, taking into account the conditions under which these countries operate and assuming that both economies face financial frictions in the investment decisions and in the provisions of private loans. The analysis also provides a positive and normative evaluation of alternative monetary policy rules.

The determination of the optimal choice of exchange rate regimes is a long standing problem in the literature. The aim of the paper is to see how this choice can be affected by financial frictions and to analyze stabilization and welfare properties of monetary rules in the transmission of external shocks in a model with capital and financial distress.

I will derive and analyze a model economy that mimics conditions in two large and financially integrated economies characterized by financial frictions<sup>2</sup> in both countries, sticky prices, imperfect competition and a rich asset structure to see to what extent policy prescriptions, such those addressed by Michael Mussa, can find consensus in a theoretical framework. The results of the paper are in accord with the Mundell's theory. Under the assumed degree of financial and trade integration between two large currency regions, and especially if credibility is not at stake, the insulation property of floating exchange rate holds. Given the presence of financial frictions the paper gives a richer explanation of the mechanism behind the stabilization properties of floating exchange rates and accounts for a difference in sign for the international transmission of shocks than that in the model without capital and financial frictions. The paper shows that in a two country model without capital the pattern of output is mainly determined by the pattern of consumption: any movement in

<sup>&</sup>lt;sup>1</sup>Speech held at Jackson Hole, Wyoming in the symposium on "Global Opportunities and Challenges".

 $<sup>^{2}</sup>$ I will consider frictions affecting investment decisions in the form of adjustment costs to capital formation and frictions in the provisions of loans in the form of agency costs.

the exchange rate under floating exchange rates determines movements in the price of an internationally traded bond. Given complete international markets, movements in the price of the international bond determine movements in consumption and consequently in output. In the model with capital and financial frictions the output mimics the movements in investment: an active monetary authority reacting to exchange rate movements generates perverse movements in the interest rate, destabilizing investment and output. The paper indicates how monetary policy can improve financial stability, stressing the importance of interest rate smoothing in tuning movements in investment and output. In the model with financial frictions the patterns of variables show a greater persistence and higher jumps on impact mostly under managed or fixed exchange rates: increasing the smoothing parameter on the lagged interest rate the impulse responses of the model with capital and financial frictions approach those of the model without capital. I evaluate the stabilization properties of different monetary regimes in terms of output and inflation volatility and I derive a welfare-based ranking of different exchange rate regimes. The welfare comparison yields a stricter result on the desiderability of the stabilization properties of floating exchange rates with respect to fixed or managed exchange rates and on the adverse effect in terms of cost of the business cycle of abandoning monetary policy discretion. In addition a systematic analysis of the cost of financial frictions under different regimes, shows that the welfare costs are higher under fixed exchange rate regimes and that they decrease when the smoothing parameter is increasing. The differential costs across regimes becomes almost the same in the model with and without financial frictions when an high interest rate smoothing is applied, showing that monetary authority can actually contrast the impact of financial frictions.

The analysis emphasizes the crucial role of monetary policy in determining the pattern of exchange rates and other macroeconomic and financial variables. A trade off will emerge between stabilization of output, inflation and prices of capital on the one side and exchange rate and trade balance on the other.

Since flexible exchange rate regimes became a reality many authors have studied the choice among exchange rate regimes. The general argument in favor of floating exchange rate regimes is its insulation property and is due to Friedman (1953) and Mundell (1960,

1961a,1961b,1963) <sup>3</sup>. Recently a considerable literature has studied the performance of different monetary policy rules in an open economy<sup>4</sup> and the comparison of exchange rate regimes in models with microfoundations. The classical mundellian theory continues to hold but these innovations yield new insights about the dynamics of model economies.

Most of the existing literature on comparison of exchange rate regimes and monetary policy rules in open economies concentrates attention on stabilization of the exchange rate<sup>5</sup> and trade balance and uses frameworks that lack markets for capital goods and frictions on investment decisions. In these models the only relevant financial asset for the economy is a state contingent bond internationally traded: any movement in the exchange rate implies movement in the price of this asset and in the wealth of consumers, with a direct effect on the overall stability of the economy. This justified the concern over a role of monetary policy in stabilizing exchange rates.

Recently the new open economy literature has examined the role on the optimal choice of exchange rate regimes of other key elements of microfounded decisions, such as pricing policies and asset market structure. The role of exchange rate pass-through has been explored by introducing into the models explicit microfoundations for the pricing decisions<sup>6</sup>. Another growing literature is exploring how the presence of borrowing constraints<sup>7</sup> and

<sup>&</sup>lt;sup>3</sup>Notice that both, Friedman and Mundell assumed that prices were fixed in the short run: this implies that a model intended to mimic that result closely should include some forms of price stickiness.

Mundell (1960, 1961a,1961b,1963) pointed out that the answer to which regime is better becomes more complicated depending on whether the source of shocks is nominal or real, on the degree of capital market imperfections, on factor mobility and on the relative size of the country.

In addition, some recent literature (Benigno P. 1999, Benigno P. and G. Benigno 2000, Gali' J. and T. Monacelli 1999) pointed out that the classical dichotomy between real and monetary shocks needs to be reinterpreted by taking into account a) an active monetary policy, b) staggered sticky prices that allow dynamic effects of monetary policy.

<sup>&</sup>lt;sup>4</sup>Ball (1998), Ghironi (19998), McCallum and Nelson (1999), Monacelli (1998), Svensson (2000), Weeparana (1998), Gali' and Monacelli (1999), Benigno G. and Benigno P. (2000), Benigno G., P. Benigno and Ghironi (2000).

 $<sup>^5\</sup>mathrm{Benigno}$  G. and P. Benigno (2000) show that under a Taylor rule the exchange rate is a non-stationary variable.

<sup>&</sup>lt;sup>6</sup>M. Devereux and C. Engel (1998), M. Devereux, C. Engel and C. Tille (1999).

This literature shows that the stabilization properties of different exchange rate regimes crucially depend on pricing policy: flexible exchange rate regimes can dominate fixed exchange rate regimes with a policy of "pricing to producer" and vice-versa for a policy of "pricing to market".

<sup>&</sup>lt;sup>7</sup>Many papers studied the role of financial frictions in closed economy: B. Bernanke and M. Gertler (1987, 1989, 1990, 1998), C. Carlstrom and T. Fuerst (1995), N. Kiyotaki and J. Moore (1997), B. Bernanke, M. Gertler and S. Gilchrist (1996, 1998), T. Cooley and K. Nam (1998), Cooley and Quadrini (1999a,

liabilities denominated in foreign currency can affect the stabilization properties of different exchange rate regimes<sup>8</sup>. This literature has mainly focused attention on small open economy and on the problem of managing currency and financial crisis.

The framework I adopt is a two country, infinite horizon model of dynamic general equilibrium. The model has several novel characteristics. I extend the set-up of the open economy described in the Galí and Monacelli (1999) by including two optimizing countries, both with a positive degree of openness, by allowing capital, adjustment costs and financial frictions in the loan markets of both countries. The design of the asset structure incorporates a domestic risk-free asset in the form of deposits in banks, a domestic and an internationally traded state contingent claim, capital and loans. The design of the financial contract links the return on deposits expressed in terms of total consumption goods with the return on capital expressed in terms of total consumption goods with the return on capital expressed in terms of trade by the degree of openness of the economy.

On this economy I perform different experiments (for technology, demand<sup>9</sup> and credibility<sup>10</sup> shocks in both countries) comparing the effects of different exchange rate regimes, implied by different monetary policy rules, on the overall stability of the economy. The simulations will be initially carried out by assuming that the two countries are symmetric in the parametrization and that the degree of openness of both countries is very low <sup>11</sup>. I will then perform several sensitivity experiments to check the robustness of the results. For a better understanding of the role played by financial distortions I will compare the model

<sup>1999</sup>b). Recently some papers have introduced financial frictions in an open economy, but attention has been concentrated basically on the small open economy where the rest of the world was not modeled and international markets were incomplete. P.Aghion, P. Bacchetta and A. Banerjee (1999, 2000) present a non microfounded version of a small open economy with financial frictions and private debt denominated in foreign currency; C. Burnside, M. Eichenbaum and S. Rebelo (2000) model a banking sector in small open economy in order to characterize financial fragility; L. Cespedes and A. Velasco (2000) and M. Gertler, S. Gilchrist and F. Natalucci (2000), E. Faia (2000) consider a microfounded version of a small open economy with a financial accelerator effect and incomplete international markets.

<sup>&</sup>lt;sup>8</sup>Aghion, Bacchetta and Banerjee 1999, 2000 show that when firms hold a large fraction of their debt in the foreign currency, monetary policy becomes increasingly complex: this is actually the case for emerging markets as it is documented by the new BIS-World Bank-IMF dataset on external debt. Even tough this fits well in the case of small open economies with very fragile financial systems and a high degree of openness, it does not seems to be crucial for two large and closed economies such as the U.S. and Europe.

<sup>&</sup>lt;sup>9</sup>The demand shock will be modeled as a shock to preferences.

<sup>&</sup>lt;sup>10</sup>The credibility shock will be modeled as shock to the exchange rate that will appear in the UIP.

<sup>&</sup>lt;sup>11</sup>The degree of openness will approximate the share of foreign consumption in the consumption index for US and Europe that is near 15 or 16 percent.

described with the two country model without capital, and with capital but in absence of financial frictions. The impulse response of the variables will show clearly the rank of the monetary regimes in terms of stabilization properties and will highlight the different transmission mechanism generated by the model with capital and financial frictions. The results of the stabilization analysis based on comparison of the patterns of variables and of the second moments of the calibrated model across different regimes will be confirmed by the results of the welfare evaluation. As a criterion of welfare evaluation of alternative monetary policy rules, I will adopt a measure of the costs of business cycles given by the fraction of non-steady state consumption that households would be willing to give up to be indifferent between a constant sequence of consumption and hours worked and the stochastic sequences of the same variables under the monetary regime considered.

Overall, the results from the model are informative and in accord with intuition. For large currency areas that are relatively closed, with low degree of openness on the trade balance.a low fraction of private debt denominated in foreign currency and with firms that use as common practice a "producer currency pricing rule", a Taylor rule with inflation and output stability goal associated to a floating exchange rate regimes can increase stability in terms of output, inflation, consumption and price of capital. The intuition of this result can be found in the analysis of transmission mechanism of the monetary policy in this context. The presence of adjustment costs on capital generates volatility in the price of capital that contributes to volatility in net worth. Borrower net worth is an important determinant of the business cycle of the economy and an essential source of volatility in output. In addition the presence of agency costs amplifies the cycle of the economy. These two elements require an active role of monetary policy: the constraint upon the monetary authority from fixed or managed exchange rates reduces the discretion of monetary policy in pursuing its active role and can generate movements in the nominal interest rate that can be perverse for the overall stability of the economy. The cost of deviating from the key goals of macroeconomic instability seems to be higher than the cost of fluctuations in the exchange rates.

The paper is organized as follow: section 2 shows the derivation of the theoretical model, section 3 describes the monetary policy rules that will be used in the comparisons, section 4 describes the calibration procedure, section 5 and 6 report the results of the simulation

experiments, section 7 shows the volatility and welfare comparison among the policy rules, section 8 contains conclusion and suggests extensions for future work. Finally section 9 provides an appendix with the derivations of the model.

### 2 The Model

There are two countries of equal size. Each country is inhabited by a continuum of agents with measure one. There is no migration across countries. Labor and capital are immobile across the two regions and there is no possibility of migration.

The population can be divided into two sets of agents: infinite lived households that supply labor, workers, and finite lived households that supply capital and entrepreneurs. Workers in both economies choose consumption and leisure, own shares in a Calvo-style sector and invest in bank deposits that pay a gross interest rate at the end of the period and in state contingent bonds denominated in domestic and foreign currency<sup>12</sup>. Entrepreneurs choose consumption, invest in capital goods and own a share of a competitive sector. In order to pay for the capital investment the entrepreneurs seek a loan while facing borrowing constraints and a default risk that can't be insured. I will assume that the consumption demand of the entrepreneurs is a negligible fraction of total consumption: their optimizing decisions will play a role only for the demand for capital and loans necessary for production in the competitive sector.

In both countries there are three sectors of firms: the wholesale producers that produce a homogenous good in a competitive fashion using capital and labor, a monopolistic sector a' la Calvo that buys the homogenous good and differentiate it little bit and a sector of capital producers.

At each point in time, each producer in the monopolistic sector is allowed to reset his price with a constant probability, independent of the time elapsed since the last adjustment. Monopolistic producers face domestic and foreign demand for their products, but do not engage in international price discrimination. The wholesale producers get capital from the entrepreneurs and produce with a constant returns to scale technology. A competitive intermediary gets money from households' deposits and offers loans to the entrepreneurs: given

<sup>&</sup>lt;sup>12</sup>Workers can then insure against the risk coming from the randomness in pricing of the firms they own.

the imperfect information the intermediary faces a monitoring problem and the solution to this generates an external finance risk premium.

#### 2.1 The Aggregate Demand Block

#### 2.1.1 Workers' Behavior in The Home Country and The Foreign Country

Workers are infinite lived agents who consume, work and hold non-monetary assets in the form of deposits in banks that pay a gross return at the end of the period and in the form of domestic and foreign state contingent bonds. Workers' utility is increasing, concave and separable over consumption and leisure<sup>13</sup>. Next, I formalize the general problem of the households in either of the two countries.

Households' utility in each country<sup>14</sup> is given by:

$$E_t \sum_{k=0}^{\infty} \beta^k [(U(C_{s,t+k}(C^H, C^F)) - V_s(N_{t+k})]$$
(1)

where  $N_t$  denotes the number of hours worked by the representative agent<sup>15</sup>,  $C^H$  is the consumption demand of households in the home country for goods produced in the home country and  $C^F$  is the consumption demand of agents of the home country for goods produced in the home country, U is increasing, concave and differentiable and V is increasing, convex and differentiable, and  $C_t$  is a Dixit-Stiglitz-Spence aggregator<sup>16</sup>.

$$C_t \equiv [(1-\gamma)^{1/\eta} C_{H,t}^{\frac{\eta-1}{\eta}} + \gamma^{\frac{1}{\eta}} C_{F,t}^{\frac{\eta-1}{\eta}}]^{\frac{\eta}{\eta-1}}$$

where  $C_{H,t}$  and  $C_{F,t}$  denotes respectively consumption of home goods and foreign goods,  $\eta$  represents the elasticity of substitution between home and foreign consumption at time t, and  $\gamma$  is the share of foreign consumption in the index and also represents the degree of openness. The CPI associated with minimum expenditure over the consumption basket is:

<sup>&</sup>lt;sup>13</sup>Notice that I am omitting money because the monetary policy will be specified in terms of the interest rate. The quantity of money will then be endogenously determined but will not play any role in determining the real side of the economy.

<sup>&</sup>lt;sup>14</sup>Since now on, the foreign variables will be denoted by a star.

<sup>&</sup>lt;sup>15</sup>I am assuming that households and firms are different agents, so I can express the households' utility in terms of leisure. In this economy agents are taking the decisions about their labor supply and firms are taking decisions about labor demand. The market clearing conditions will give the amount of labor in the economy.

<sup>&</sup>lt;sup>16</sup>The quantity of the composite consumption good is given by:

Households solve a three stage decision problem: they choose the optimal allocation of each variety in the first stage, the optimal allocation between home and foreign consumption and in the third stage they make consumption, labor and investment decision by maximizing the utility subject to a sequence of budget constraints for each state of the world. Appendix 9.1.1 contains the derivation of the maximization problem, the loglinearized expressions of the aggregate demand relations and the derivations of some identities for a two country economy.

By using the optimal consumption demands derived in the first two stages of the maximization problem I am able to write the budget constraint in each state in a compact form:

$$\int_{0}^{1} [P_{H,t}(i)C_{H,t}(i) + P_{F,t}(i)C_{F,t}(i)]di + E_t \{d_{t,t+1}B_{t+1}\} + E_t \{e_t d_{t,t+1}^*B_{t+1}^*\} + D_{t+1} \le (2)$$

$$\leq W_t N_t + T_t + B_t + B_t^* + R_t D_t$$

where  $B_t$  is the quantity of the domestic bond<sup>17</sup>,  $B_{t+1}^*$  is the quantity of the international bond and  $E\{d_{t,t+1}\}$  is the expected price of one unit of the domestic bond,  $E\{e_t d_{t,t+1}^*\}$  is the expected price of one unit of the international bond and  $e_t$  is the nominal exchange rate,  $D_{t+1}$ is the quantity of bank deposits and  $R_t^n$  is the nominal gross return on deposits ( $R_t^n = R_t \frac{P_t}{P_{t+1}}$ where  $R_t$  is the real gross return expressed in terms of CPI consumption goods),  $W_t$  is the nominal wage and  $T_t$  are the nominal lump-sum transfers (or taxes). All the variables are expressed in terms of the domestic currency.

$$P_t \equiv [(1-\gamma)P_{H,t}^{1-\eta} + \gamma P_{F,t}^{1-\eta}]^{\frac{1}{1-\eta}}$$

The indices for home and foreign consumption are given by Dixit-Stiglitz aggregators over a continuum of goods with the property of constant elasticity of substitution over time:

$$C_{s,t} = \left(\int_0^1 C_{s,t}(i)^{\frac{\varepsilon-1}{\varepsilon}} di\right)^{\frac{\varepsilon}{\varepsilon-1}}$$

where  $\varepsilon$  denotes the intertemporal elasticity of substitution.

 $^{17}{\rm The}$  return paid by the contingent bond is actually dependent on the state of nature, but for notational simplicity I avoid this cumbersome notation.

Combining the results on optimal allocation of expenditures from the first two stage of the maximization, I get the total demand for good i:

$$Y_{t+k}^d(i) = \left(\frac{P_{H,t}(i)}{P_{H,t+k}}\right)^{-\varepsilon} (C_{H,t+k} + C_{H,t+k}^*)$$
(3)

where  $C^*_{H,t+k}$  is the foreign demand for goods produced in the home country. A similar expression holds for the foreign country.

As a result of the third stage maximization problem I get the following optimality conditions:

$$\beta \frac{U_c(C_{t+1})}{U_c(C_t)} (\frac{P_t}{P_{t+1}}) = d_{t,t+1}; \beta \frac{U_c(C_{t+1})}{U_c(C_t)} (\frac{P_t}{P_{t+1}}) (\frac{e_{t+1}}{e_t}) = d_{t,t+1}^*$$
(4)

$$U_c(C_t)W_t/P_t = -V_N(N_t)$$
(5)

$$R_t^n = E_t \{ d_{t,t+1} \}^{-1}; R_t^{n*} = E_t \{ d_{t,t+1}^* \}^{-1}$$
(6)

$$E_t \{ C_{t+1}^{-\sigma} \frac{P_t}{P_{t+1}} [R_t^n - R_t^{n*} (\frac{e_{t+1}}{e_t})] \} = 0.$$
(7)

Equations (4) are the FOC with respect to the domestic and international contigent bond in each state of the world. Equation (5) gives the labor supply. Equation (6) are arbitrage conditions between the state contingent bonds and the deposits<sup>18</sup>. Finally equations (7) is a version of the UIP and is obtained by equating the conditional expectations of the first two and using (6).

I assume that in both the home and the foreign country there is a complete set of state contingent bonds in the Arrow-Debreu sense This implies that for the consumers in the other country we get a similar Euler equation:

$$U'(C_t^*) = R_{t+1}^n \beta U'(C_{t+1}^*) (\frac{P_t^*}{P_{t+1}^*}) (\frac{e_t}{e_{t+1}}).$$
(8)

<sup>&</sup>lt;sup>18</sup>They are obtained from equating the FOC with respect to the deposits with the conditional expectation of the FOC with respect to the contingent bonds.

### 2.2 The Aggregate Supply Block: Firms' Behavior At Home and In The Foreign Country

I will solve the firm's maximization problem for the home country. A similar derivation holds for the firms in the foreign country.

#### 2.2.1 The Competitive Sector and The Capital Producers

There is a competitive sector owned by finite lived agents, the entrepreneurs, <sup>19</sup> (firms have an exogenous probability of failure  $\zeta$ ). The sector produces a homogenous good, hiring capital and labor. There is a continuum of firms indexed by k. Each firm is subject to idiosyncratic shocks and to aggregate shocks on the return on capital. At the beginning of each period the entrepreneur observes the aggregate shock. Before buying capital the entrepreneur goes to the loan markets and borrows money from the intermediary by making a contract which is written before the idiosyncratic shock is recognized. With the money borrowed from the intermediary, the entrepreneur goes to the factor market to hire capital. The optimal demand for capital (and labor) is determined by minimization of total production cost subject to a constant return to scale production function<sup>20</sup>. After investing in capital goods the entrepreneur observes the idiosyncratic shock.

For the whole competitive sector there is an aggregate demand for capital and labor. The aggregate demand for capital will give the return on capital  $(R^k)$  for each realization of

<sup>20</sup>The problem is a standard one and looks like this. Each competitive firm k chooses  $K_t(k)$  and  $L_t(k)$  to:

$$Min\frac{W_t}{P_t}L_t(k) + Z_t(k)K_t$$

s.t.

$$Y_t(k) = A_t L_t(k)^{\alpha} K_t(k)^{1-\alpha}$$

Both, the production function and the labor supply are increasing, concave, constant return to scale and differentiable. Setting the Lagrangian and taking derivatives with respect to  $K_t(k)$  and  $L_t(k)$  we get:

$$\frac{1}{MC_t}\frac{W_t}{P_t} = (1-\alpha)\frac{Y_t(k)}{L_t(k)}; \frac{1}{MC_t}MPK_t = \alpha\frac{Y_t(k)}{K_t(k)}$$

where  $MC_t$  is the real marginal cost evaluated at the home currency.

<sup>&</sup>lt;sup>19</sup>This assumption is necessary to rule out the possibility that the borrowing constraint will not be binding. If the firm has an infinite life or an expected survival time sufficiently long, it can accumulate a large amount of assets and ease the borrowing constraints.

the idiosyncratic shock. In equilibrium the expected real return on capital is:

$$E(R_{t+1}^k) = E(\frac{mc_t \alpha \frac{Y_t(k)}{K_t(k)} + Q_{t+1}(1-\delta)}{Q_t})$$
(9)

 $mc_t$  is the real marginal cost given by  $\frac{1}{mc_t} \frac{MPK_t}{P_{H,t}} = \alpha \frac{Y_t(k)}{K_t(k)}$ ,  $Y_t(k) = A_t N_t(k)^{\alpha} K_t(k)^{1-\alpha}$ ,  $A_t$  is the technology shock,  $N_t(k)$  is the labor input,  $K_t(k)$  is capital,  $Q_t$  is the real price of capital and  $\delta$  is the depreciation rate. Given that capital is not traded internationally, every variable is expressed in terms of domestic consumption good.

After determining the demand for capital, entrepreneurs use the money borrowed from the intermediary to buy capital. The sector of capital producers faces adjustment costs and sets a price for capital as a function of  $R^{k_{21}}$ . From the maximization problem of the capital producers<sup>22</sup> we get the real price of capital:

$$Q_t = [\phi'(\frac{I_t}{K_t})]^{-1}$$
(10)

where  $I_t$  is investment. The law of motion of capital is:

$$K_{t+1} = \phi(\frac{I_t + I_t(z)}{K_t})K_t + (1 - \delta)K_t$$
(11)

where  $\phi(\frac{I_t+I_t(z)}{K_t})$  is a production function that embeds adjustment costs, where  $I_t(z)$  is the aggregate investment from all the other firms.

<sup>22</sup>The introduction of adjustment costs will follow Kiyotaki and Moore (1998). There is a competitive sector for firms producing capital. These firms operate with a technology embedding external adjustment costs. The adjustment costs are modeled trough the following production function for capital:  $\phi(\frac{I_t+I_t(z)}{K_t})K_t$  where  $\phi(.)$  is increasing and concave and  $\phi(0) = 0$  and  $I_t$  is the aggregate level of investment and  $I_t + I_t(z)$  is the gross output of new capital goods. The problem of the representative firm is to choose  $I_t(z)$  to:

$$MaxQ_t\phi(\frac{I_t + I_t(z)}{K_t})K_t - I_t(z)$$

subject to

$$K_{t+1} = \phi(\frac{I_t + I_t(z)}{K_t})K_t + (1 - \delta)K_t$$

having normalized the price of  $I_t(z)$  equal to one.

 $<sup>^{21}</sup>$ The presence of a third sector producing capital that faces adjustment costs permits derivation of a price for capital in the form of Tobin's q. An alternative way of modeling this would be to assume that the competitive sector faces a dynamic decision on capital investment. Firms get a return given by the marginal product of capital by using capital goods as production input in each period and they get a return in the form of capital gains by selling capital from one period to the other. This second alternative is quite cumbersome.

#### 2.2.2 The Behavior of the Monopolistic Sector

The second production sector in the economy has the task of differentiating the homogenous good: it is a monopolistic competitive sector. In choosing the optimal price they optimize in the Calvo fashion. The optimizing behavior of this sector will provide the pricing function for the final good and through this it will provide a relationship that works as a Phillips curve.

Agents are monopolists and prices are staggered. In each period the agent faces a fixed probability of adjusting prices  $(1 - \vartheta)$ . In this event the agent chooses the price  $p_{s,t}(i)$  with s = H, F (H stands for home country and F stands for foreign country) for the produced differentiated good i so as to maximize the expected utility resulting from sale revenues minus nominal marginal costs in each of the future states in which the price commitment still applies. The maximization (Appendix 9.2) is performed taking as given  $P, P_H, P_F$  and Cand subject to the aggregate demand curve(3). The solution to the maximization problem is:

$$P_{s,t}^{new} = \mu \frac{E_t \sum_{k=0}^{\infty} (\vartheta)^k \Lambda_{t,t+k} Y_{t+k}(i) M C_{t+k}}{E_t \sum_{k=0}^{\infty} (\vartheta)^k \Lambda_{t,t+k} Y_{t+k}(i)}$$
(12)

where  $\mu$  is a mark-up,  $\vartheta$  is the probability that the price is fixed in each period,  $Y_{t+k}(i) = \left(\frac{P_{H,t}(i)}{P_{H,t+k}}\right)^{-\varepsilon} (C_{H,t+k} + C^*_{H,t+k})$  and  $\Lambda_{t,t+k} = \beta^k \left(\frac{C_t}{C_{t+k}}\right) \left(\frac{P_t}{P_{t+k}}\right)$ . The new price is determined as a constant mark-up over the discounted future stream of marginal costs. Embedded in the maximization problem of the monopolistic sector is the assumption that the producers set the price of their goods in domestic currency. The price of that good in the foreign market is then determined in accord with the prevailing exchange rate<sup>23</sup>.

 $<sup>^{23}</sup>$ This assumption is certainly true for the US producers, but it may not be true for producers of a small country: given the size of the US market, foreign producers may find it convenient to set the price in dollars in order to avoid the risk of exchange rate fluctuations. Notice that the model economy is build up as a two region world with two big countries, such as the US. and Europe, that have a low degree of openness (the share of domestically produced goods that are sold outside is around 15 or 16 percent for both countries). This allows us to consider as reasonable the assumption that also European producers set prices in Euro. For this reason the microfoundations of the model do not incorporate producers who discriminate between markets and I will, in general, assume "producer currency pricing" for each region.

#### 2.3 The Financial Intermediary

The financial intermediary collects money from deposits and supplies loans to the entrepreneurs facing an incentive problem due to asymmetric information. The asymmetric information in this economy arises from the fact that firms observe the idiosyncratic shock but households and banks can do so only at some cost. The role of financial intermediary is to minimize the monitoring cost by pooling projects<sup>24</sup>. The incentive problem is solved via the design of a contract that takes the form of costly state verification that is shown to be an optimal debt contract in Gale and Hellwig (1985)<sup>25</sup>. I will introduce financial frictions in the general equilibrium following a standard strategy like that in Bernanke, Gertler and Gilchrist (1998) or Cooley and Quadrini (1999). Given that the fraction of private loans in foreign currency for the Euro area and the U.S. economy is negligible, I will assume that the entire loan demand is expressed in domestic currency.

Appendix 9.3 shows the maximization problem faced by the intermediary. The solution to the problem yields the following loglinearized expression for the external finance premium:

$$\frac{E_t(R_{t+1}^k)}{R_t(P_t/P_{H,t})} = -F[\frac{\Pi_t}{Q_t K_t}]$$
(13)

where  $\Pi_{t+1}$  are net worth of the entrepreneurs, and  $\frac{E_t(R_{t+1}^k)}{R_t(P_t/P_{H,t})}$  is the external finance premium and F is the elasticity of the risk premium to the collateral. Both variables are expressed in terms of domestic consumption goods. Equation (13) entails a negative relation between the finance premium and the collateral of the firms, expressed by the ration between net worth and capital.

The aggregate level of net worth ( $\Pi$ ) is derived by assuming that the change in the wealth of the entrepreneur is given by the return on capital investment minus the cost of loans and that only a fraction  $\xi$  of entrepreneurs remain alive period by period:

 $<sup>^{24}</sup>$ There is no asymmetric information between households and the intermediary: the bank pays a fixed rate of return on deposits.

 $<sup>^{25}</sup>$ A standard debt contract requires a fixed payments in the states for which the firm is solvent and requires the intermediary to recoup as much as possible of the debt when the firm cannot meet the payment.

$$\Pi_{t+1} = \xi [R_t^k Q_{t-1} K_t - (1+\varkappa)(1+R_{t-1}\frac{P_t}{P_{H,t}})(Q_{t-1} K_t - \Pi_t)]$$
(14)

where  $(1 + \varkappa)$  is the expost cost of loans.

Notice that the return on deposit is evaluated in terms total consumption goods while the return on capital is evaluated in terms of domestic consumption goods: this generates a loglinearized expression for the external finance premium  $(E_t(\hat{r}_{t+1}^k) - \hat{r}_t = \gamma \hat{s}_t - v[\hat{\Pi}_{t+1} - (\hat{q}_t + \hat{k}_{t+1})])$  that depends positively on the terms of trade and whose dependence is defined by the degree of openness. An increase in the terms of trade due to an increase in the domestic prices implies an increase in the return of capital goods and an increase in the external finance risk premium.

#### 2.4 Fiscal Authority

For simplicity I am assuming that the fiscal authority in each country does not make public expenditures. The fiscal authority just makes lump sum transfers among individuals. The budget constraint at date t in the home country (the same holds for the foreign country) is:

$$\int_{0}^{1} T_{s}(i)di = 0.$$
(15)

#### 2.5 The Equilibrium Conditions

In equilibrium the market clearing condition implies:

$$Y_t(i) = C_{H,t}(i) + C_{H,t}^*(i) + I_{H,t}(i).$$
(16)

Given certain assumption on the aggregator functions<sup>26</sup> for each country the following

$$I_{s,t} = \left(\int_0^1 I_{s,t}(i)^{\frac{\varepsilon-1}{\varepsilon}} di\right)^{\frac{\varepsilon}{\varepsilon-1}},$$

<sup>&</sup>lt;sup>26</sup>The aggregation problem has been solved by assuming that the aggregate investment in both countries can be represented trough a CES aggregator:

and that aggregate outputs can be approximated by the sum of individuals output at least in a neighborhood of the steady state. There is no trade on investment goods, meaning that each country uses its own production of capital goods as input.

equilibrium conditions must hold:

$$Y_t = C_{H,t} + C_{H,t}^* + I_{H,t}; \ Y_t^* = C_{F,t} + C_{F,t}^* + I_{F,t}.$$
(17)

Market clearing condition for bonds denominated in the currencies of the two countries requires these asset to be in zero net supply:

$$B_{H,t} + B_{F,t}^* = 0; \ B_{F,t} + B_{H,t}^* = 0.$$
(18)

Finally the real demand for loan has to be equal to the real supply of loans:

$$\frac{D_t}{P_t} = L_t = (Q_t K_t - \Pi_t); \frac{D_t^*}{P_t^*} = L_t^* = (Q_t^* K_t^* - \Pi_t^*).$$
(19)

#### 2.6 The Loglinearized version of the model

I now present the complete macroeconomic framework. Lower case variables denote percentage deviations from the steady state (Appendix 9.4). I will present the equations for the home country, but those for the foreign country look the same.

• Aggregate Demand.

$$\widehat{y}_{H,t} = (\zeta_h - \zeta_{h^*})(\eta(1-\gamma)\overset{\wedge}{s_t}) + \zeta_h \widehat{c}_t + \zeta_{h^*} \widehat{c}_t^* + \zeta_{I_h} \widehat{i}_{H,t}$$
(20)

$$\hat{c}_{t} = E_{t}\{\hat{c}_{t+1}\} - \frac{1}{\sigma}(\hat{r}_{t}^{n} - E_{t}\{\pi_{H,t+1}\}) + \frac{\gamma}{\sigma}E_{t}\{\Delta \hat{s}_{t+1}\} - \frac{1}{\sigma}(\nu_{t+1} - \nu_{t})$$
(21)

$$E_t(\hat{r}_{t+1}^k) - \hat{r}_t - \gamma \hat{s}_t = -F[\Pi_{t+1} - (\hat{q}_t + \hat{k}_{t+1})]$$
(22)

$$\hat{r}_{t+1}^{k} = (1-g)(\hat{y}_{H,t+1} - \hat{k}_{t+1} + \hat{m}c_{t+1}) + g(\hat{q}_{t+1} - \hat{q}_{t})$$
(23)

$$\hat{q}_t = \varphi(\hat{i}_t - \hat{k}_t) \tag{24}$$

$$\overset{\wedge}{s_t} = (\overset{\wedge}{r_t} - E_t\{\pi_{F,t+1}^*\}) - (\overset{\wedge}{r_t} - E_t\{\pi_{H,t+1}\}) + E_t\{\overset{\wedge}{s_{t+1}}\} + f_t$$
(25)

In addition I assumed that the consumption basket can be represented by a CES aggregator and that aggregate output can be represented as the sum of the individuals outputs:

$$Y_{s,t} = \left(\int_0^1 Y_{s,t}(i)^{\frac{\varepsilon-1}{\varepsilon}} di\right)^{\frac{\varepsilon}{\varepsilon-1}}$$

where s = H, F.

• Aggregate Supply Block.

$$\hat{y}_{H,t} = \hat{a}_t + \alpha \hat{k}_t + (1 - \alpha) \hat{n}_t$$
(26)

$$\hat{y}_{H,t} + \hat{mc}_t - \sigma \hat{c}_t = (1+\tau) \hat{n}_t + (1-\gamma) \hat{s}_t$$
(27)

$$\pi_{H,t} = \beta E_t(\pi_{H,t+1}) + \lambda(\stackrel{\wedge}{mc_t}) \tag{28}$$

• Law of Motion for State Variables.

$$\hat{\vec{k}}_{t+1} = \delta \hat{\vec{i}}_t + (1-\delta)\hat{\vec{k}}_t \tag{29}$$

$$\stackrel{\wedge}{\Pi}_{t+1} = \frac{\xi RK}{\Pi} (\stackrel{\wedge}{r}_t^k - \stackrel{\wedge}{r}_t) + \stackrel{\wedge}{r}_t - \gamma \stackrel{\wedge}{s}_t + \stackrel{\wedge}{\Pi}_t + \phi^n \tag{30}$$

• Evolution of Processes for the Stochastic Variables (shock to technology, preferences and exchange rates):

$$\hat{a}_t = \rho_a \hat{a}_{t-1} + \varepsilon_t^a; \hat{\nu}_t = \rho_\nu \hat{\nu}_{t-1} + \varepsilon_t^\nu; \hat{f}_t = \rho_\nu \hat{f}_{t-1} + \varepsilon_t^f$$
(31)

• where 
$$\zeta_h = (1 - \gamma) [1 - \frac{\delta \alpha}{\mu(R^k - 1 + \delta)}], \ \zeta_{h^*} = \frac{\gamma}{1 - \gamma} \zeta_h, \zeta_I = \frac{\delta \alpha}{\mu(R^k - 1 + \delta)}, \ g = (1 - \delta) / [(1 - \delta) + \alpha Y/K], \ \nu = \psi(R^k/R) / \psi'(R^k/R), \ \varphi = [(\phi(I/K)^{-1})'(I/K) / (\phi(I/K)^{-1})''], \ \lambda = \frac{(1 - \vartheta)(1 - \beta \vartheta)}{\vartheta}, \ \delta = \phi(\frac{I_t}{K_t}) = \frac{I_t}{K_t}, \ \phi^n = \frac{(R^k/R - 1)K}{\Pi} (r_t^k + q_{t-1} + k_t), E(a) = 0, \ Var(a) = 1.$$

Equation (20) is obtained by substituting in the loglinearized version of the resource constraint the demand for domestic and foreign consumption good. Equation (21) is the loglinear Euler equation after substituting the expression for the CPI domestic inflation. Equation (22) is the loglinear external finance risk premium. Equation (23) is the loglinear expected return on capital. Equation (24) is the loglinear Tobin's q. Equation (25) is the loglinear UIP expressed in real terms. Equation (26) is the loglinear production function of the competitive sector. Equation (27) is obtained by loglinearizing the equilibrium condition for the labor market. Equation (28) is the Phillips curve. For the foreign country we have the same set of equations.

### 3 The Monetary Policy Rules

There is an active monetary policy: the monetary authority sets the short term nominal interest rate by reacting to endogenous variables. Each monetary and exchange rate regime will be identified with a different specification of the monetary policy rule.

The comparison between different specifications of the monetary rule and the exchange rate regime at the same time seeks for the scenario that yields the greatest degree of stability. I will consider the general class of the Taylor rules of the following form:

$$\stackrel{\wedge^n}{r} = \varpi_{\pi}(\pi_{H,t-l} - \bar{\pi}) + \varpi_y(\stackrel{\wedge}{y}_{t-l} - \bar{y}) + + \varpi_e(\stackrel{\wedge}{e}_t - \bar{e}) + \stackrel{\wedge}{m}_t \tag{32}$$

where  $\varpi_{\pi}$  is the weight the monetary authority puts on the deviation of inflation from the target  $(\bar{\pi}), \varpi_y$  is the weight that the monetary authority puts on the deviation of output from the target level  $(\bar{y}), \varpi_e$  is the weight that the monetary authority puts on the deviation of the exchange rate from the target  $(\stackrel{\wedge}{e_t} - \bar{e})$  and  $\stackrel{\wedge}{m_t}$  is a monetary policy shock that evolves according to  $\stackrel{\wedge}{m_t} = \rho_m \stackrel{\wedge}{m_{t-1}} + \varepsilon_t^m$ .

Following Clarida-Gali'-Gertler (1998) and Rotemberg-Woodford (1998) I assume that monetary policy goal is to smooth changes in interest rate:

$$\hat{r}_{t}^{n} = (1-\chi)\hat{r}_{t}^{n} + \chi \hat{r}_{t-1}^{n}$$
(33)

I will identify a regime of pure floating exchange rate with a Taylor rule of the form (32) in which  $\varpi^e = 0$ . I will consider two different kind of managed exchange rate systems: one identified by a Taylor rule of the form (32) in which  $\varpi^e > 0$  and one identified by a Taylor rule in which the monetary authority sets a positive reaction coefficient on the change of the exchange rate over time  $(e_t - e_{t-1})$ . Finally I will identify a regime of pure peg with the following monetary rule<sup>27</sup>:

$$\stackrel{\wedge^n}{r_t} = \stackrel{\wedge^{n*}}{r_t}.$$
(34)

 $<sup>^{27}{\</sup>rm This}$  rule follows just from the assumption that the monetary authority sets equal to zero the deviation of the exchange rate from its steady state value.

I will also compare the situation in which the specification is perfectly symmetric for both and the situation in which the "leader" country uses a different parametrization and a specification different from that of the "follower".

### 4 Calibration

The model is parametrized as followed. The two country are assumed to be symmetric in preference and technology specifications in the first simulations. Time is taken to be measured in quarters.

**Preferences:** I set the discount factor  $\beta = 0.99$ , so that the annual interest rate is equal to 4 percent. As in most of the literature on RBC, I set the elasticity of substitution between domestic and foreign goods  $\eta$  equal to 1.5. The parameters on consumption and labor in the utility function are set equal to one to generate a log utility and a unity supply of labor<sup>28</sup>. I set the degree of openness at  $\gamma = 0.15$  (and I allow for variations ranging between 0.10 and 0.20) that is consistent with the US-Europe case. The intertemporal elasticity of substitution of labor is  $\phi = 1$ , and the intertemporal elasticity of substitution is  $\sigma = 1^{29}$ .

**Technology:** the share of capital in the production functions  $\alpha = 0.3$ , the quarterly depreciation rate  $\delta = 0.025$ , the steady state mark-up value  $\mu = 1.2$ . The probability of adjusting prices in each period  $\vartheta$  is set equal to 0.75, a value consistent with an average period of one year between price adjustment. The elasticity of the price of capital with respect to investment output ratio  $\varphi = 0.5$ .

Financial frictions parameters: The elasticity of the premium with respect to the leverage ratio is F = 0.05, the probability that a firm will be alive next period is  $\xi = 0.975$ . The steady state ratio of net worth over capital is 0.5. The values calibrated as described are the result of assuming a risk premium in steady state of 200 hundred basis point, a business failure rate  $(F(\bar{\omega}))$  of three percent, a ratio of capital to net worth of 2 and a ratio of debt to capital of eighty percent. All the values are consistent with data in industrialized countries, like U.S. and Europe.

<sup>&</sup>lt;sup>28</sup>These values are compatible with those of a steady state balanced growth.

 $<sup>^{29}{\</sup>rm The}$  intertemporal elasticity of substitution also gives the elasticity of the aggregate exports demand with unit real exchange rate.

Monetary policy parameters: I fix the weight on inflation in the Taylor rule at  $\varpi_{\pi} = 1.5$  and the smoothing parameter at  $\chi = 0$  or  $\chi = 0.9$ . The other parameters will vary according to the experiment performed.

Exogenous shocks: The persistence of the shocks varies between 0.8 and 0.9.

The equilibrium of the model is characterized as the solution of the system of expectation difference equations of the loglinearized form<sup>30</sup>.

### 5 Fixed Versus Floating Exchange Rates

This section describes the dynamics of the home economy driven by stochastic innovations in the technology process and in the monetary policy process in both the home and foreign country and compares the impact of these shocks under floating and pure peg exchange rate regimes. I will assume for simplicity that the monetary authority of the foreign country is following a Taylor rule of the form described in (32). For the home country I will compare different regimes: pure floating, managed and pure peg<sup>31</sup>.

Productivity Shock in the Home Country. Figure 1 shows the impulse response functions of the home economy in the model with capital and financial frictions to a 1% technology shock (with a 0.9 autoregressive coefficient) in the home country under both, floating (solid curves) and fixed (dashed curves) exchange rate regimes with a zero smoothing parameter factor in the Taylor rule. The persistence of patterns of the variables under fixed exchange rates looks greater than that under floating exchange rates. For all the variables, under fixed exchange rate regimes, the jump on impact is bigger and the convergence to the steady state

$$E_t \sum_{i=-m}^{n} A_i X_{t+i} = 0, t \ge 0$$

<sup>&</sup>lt;sup>30</sup>The loglinearized system can be described by a general homogenous matrix equation:

where m is the number of leads, n is the number of lags,  $A_i$  are the structural coefficient matrices, and  $A_n(n = 1)$  is not full-rank. I apply the solution method developed by Anderson and Moore (1985) which enables us to deal with possibly singular systems, unlike the Blanchard-Khan (1980).

 $<sup>^{31}</sup>$ For Europe I will set the coefficient on the income deviations from the steady state in the Taylor rule always equal to 0. The EU Treaty unambiguously assigns to the ECB the mandate of maintaining price stability. On the contrary the statutory mandate of the Fed is less clearly defined and US monetary policy is usually interpreted to be sensitive to both output and inflation. For the U.S. I will let this weight to move from 0 to 0.5.

takes a longer time. The positive effect of a technology shock causes a real and nominal depreciation. Under fixed exchange rates the nominal interest rate goes up as a result of the active role of monetary policy: this induces a decrease in investments, net worth, price of capital and output. The decrease in inflation leads to a decrease in the terms of trade. Notice that output follows a pattern pretty similar in the shape to that of investment and net worth. The nominal exchange rate and, consequently, the price of the international traded bond is much more unstable under floating exchange rates, while the price of capital looks much more unstable under fixed exchange rates.

When I shut off the risk premium the impulse responses look pretty similar to the ones in the model with agency costs: with a home technology shock under fixed exchange rates the destabilization effect due to agency costs is not significant. On the contrary, as I will explain later, under a foreign technology shock the acceleration effect is very strong. The explanation of the difference in the acceleration effect between the two shock can be mainly found in the following intuition: a positive technology shock in the home country should imply a fall in the nominal and consequently in the real interest rate that should boost the economy, although this does not happen under fixed exchange rates because the interest rate is pegged to the world interest rate<sup>32</sup>. On the other side with a foreign technology shock there is a direct impact on the interest rate that can't be controlled by a monetary authority subject to external constraints.

Figure 2 shows the pattern of the variables in the same experiment in the model without capital and financial frictions. In this case output looks more destabilized under floating exchange rates than under fixed and follows a pattern pretty close to the pattern of consumption. In the previous model capital and net worth were key determinants of output: movements in the interest rate due to the active monetary authority had a destabilizing effect on capital and consequently on output. In the model without capital, consumption is a key determinant of output: movements of nominal exchange rates determine movements in the price of the international bond, destabilizing consumption and consequently output. Notice that even the sign of the transmission of shocks under fixed exchange rates is dif-

 $<sup>^{32}</sup>$ Faia E. and Monacelli T. (2001) show that in a small open economy with positive degree of financial exposure (a positive fraction of debt denominated in foreign currency) under a home technology shock fixed exchange rates can work as a stabilizer compared to flexible exchange rates.

ferent in the model with capital from that in the model without capital: negative in the first one and positive in the second. In the setup with capital and with financial frictions in both countries a foreign monetary authority following an interest rate rule that targets endogenous variables such as output gap and inflation, should pursue a stronger active role. In response to a positive technological shock in the home country the central bank reacts by increasing the interest rate more than in the model without capital. Under pure fixed exchange rates the foreign interest rate provides an anchor inducing similar movements in the interest rate of the home country. The strong increase in the interest rate in the home country depresses output and consumption.

Figure 3 shows the impulse response of the variables in the model with capital and financial frictions when the monetary authority adopts a smoothing parameter factor of 0.8/0.9 on the lagged interest rate. The pattern of variables looks more stable than in the previous case (Figure 1) and comes closer to the pattern of variables in the model without capital. This shows that the monetary authority can improve financial stability by smoothing the interest rate even without reacting to asset price variability: the destabilization effect of a variable price of capital and the presence of an external finance premium can be mitigated by tuning the interest rate closely.

Productivity Shock in the Foreign Country. Figure 4 shows the impulse response functions of the home country variables in a model with capital and agency costs to a 1 percent shock in the technology process (with an autoregressive parameter of 0.9) for the foreign country under both floating and fixed exchange rate regimes and a smoothing parameter of 0 on the interest rate. Under fixed exchange rate regimes almost all macroeconomic variables, apart from exchange rate and trade balance show, a higher persistence and volatility. In order to match a decline of the interest rate abroad the output in the home country shows a big expansion that also leads to an increase in inflation. Due to the expansion in inflation there is a decline in the terms of trade that depress the trade balance. The increase in output and the decrease in the nominal interest rate produce an increase in net worth and in the price of capital. Under floating exchange rates all the variables, apart from nominal exchange rate, terms of trade and trade balance, show a low degree of persistence.

I have repeated the same experiment in the model with capital by assuming a zero

external finance risk premium (Figure 5): the impulse responses are much more stable. The jump on impact of output is 2, while in the model with a positive external finance risk premium is 4. Inflation declines on impact and the persistence in the movements of the terms of trade declines. When I shut off the risk premium the economy looks much more stabilized mostly under fixed exchange rates, while there is no such big difference in the impulse responses under floating exchange rates. Finally one should notice that as long as we increase the interest rate smoothing the difference in the impulse responses between the two models, with and without agency costs, decreases.

A comparison with the model without capital shows that the insulation property of the floating exchange rate looks even stronger in a model with variable price of capital and agency costs. Figure 6 shows the impulse response of the model without capital: the output has the same shape under both regimes but the jump on impact is much smaller. Under a floating exchange rate inflation looks more stable in the model with capital than in the model without capital. Finally one can notice an increase of the terms of trade in the model with capital and a decrease in the model without capital.

Again repeating the same experiment in the model with capital and agency costs assuming a smoothing parameter factor of 0.8/0.9 (Figure 7) shows that the economy approaches the economy without capital in terms of stability: the jump on impact of both, output and inflation, declines. We can recognize a monotonic relation between the coefficient of interest rate smoothing and the volatility of output: when the coefficient increase given the presence of financial frictions the volatility of output decreases.

Monetary Shock in the Foreign Country, Demand Shock and Credibility Shock. The mundellian theory recognizes a dichotomy between real shocks, that affects the interest rate, and nominal shocks, identified as shocks to money demand. With real shocks the insulation property of floating exchange rates holds, while with nominal shocks movements in the exchange rates can destabilize the entire economy. In a model where money is endogenously determined a monetary policy shock is identified as a random process that enters the monetary policy rules. For this reason its impact on the economy will look very much like the impact of a real shock. In this sense the classical dichotomy should be reinterpreted. A 1 percent shock to the monetary policy of the foreign country generates impulse responses of the home country variables that are very similar to those generated by a 1 percent foreign technology shock. When the process for the monetary shock is autoregressive instead than i.i.d. the destabilizing property of the fixed exchange rates is even stronger than in the case of the foreign technology shock.

I perform simulations of the model with both, a home and a foreign shock to preferences<sup>33</sup>: this will impact consumption demand and will act as a demand shock. In both cases still fixed exchange rates look more destabilizing. A negative demand shock in the home country generates, under fixed exchange rates, a recession with big destabilizing effect. A negative demand shock in the foreign country has a positive impact in the home country: the reduction of the foreign demand, induces a reduction in the foreign interest rate and trough the uncovered interest parity it also induces a reduction in the home interest rate. The fall in the home interest rate generates a positive acceleration effect to the home economy.

Finally I perform simulation of the model under credibility shocks<sup>34</sup> (shocks to the exchange rates in the UIP). Even in this case fixed exchange rate regimes are more destabilizing. Under fixed exchange rates, a positive shock to the expectation of devaluation generates an increase of the home interest rate and depresses the economy.

# 6 Managed Exchange Rates Versus Floating and Versus Fixed Exchange Rates

The purpose of this section is the comparison of managed exchange rate regimes with floating and with fixed exchange rate regimes. For simplicity I will assume that only the monetary authority of the home country is reacting to the exchange rate.

*Productivity shock in the home country and abroad.* Figure 8 and 9 shows the impulse response to a home and foreign technology shock, under 0.8 persistence of the shock, of the model with capital and financial frictions under floating, fixed and two kind of managed exchange rates, one pegging the level and the other pegging the change in the exchange rate<sup>35</sup>.

<sup>&</sup>lt;sup>33</sup>To see how the shock enters the loglinear demand for consumption see equations 47 and 48 in Appendix 9.1.

 $<sup>^{34}</sup>$ To see how the shock enters the UIP see equation 53 in Appendix 9.1.

 $<sup>^{35}</sup>$ I assume as benchmark a weight on the exchange rate of 0.5. I then let the weight vary to check robustness.

There appears to be a monotone ranking of the three regimes in terms of stabilization: floating exchange rate regimes look more stabilizing than managed regimes and managed regimes look more stabilizing than fixed exchange rate regimes in terms of output and inflation persistence. Management of the change in the nominal exchange rate yields more stabilization than managements of the level and leads to impulse response functions that are closer to those with floating exchange rates. Given that the exchange rate is a non-stationary variable a reaction to the change in levels helps the monetary authority to get rid the non-stationary elements.

We can conclude that management of the level of the exchange rates delivers almost the same destabilization effect than a system of pure fixed exchange rate appearing equally undesirable for the monetary authority.

# 7 Volatility of Variables under Different Regimes and Welfare Comparison

Most of the positive analysis of this paper has centered on comparison of the time paths that the variables would follow under different regimes and on comparison among the different degree of persistence shown by the variables in their response to a shock. The purpose of this section is to evaluate from a normative point of view the behavior of different monetary policies in an open economy and in presence of financial frictions. I will first evaluate the stabilization properties of the different regimes by comparing the volatilities of the variables<sup>36</sup>. I compute the second moments of the calibrated model.

Table 1 and 2 show the second moment<sup>37</sup> of the model with capital, financial frictions and in absence of interest rate smoothing under home and foreign technology shock. Fixed exchange rates appear to be the more destabilizing than any other regime even in terms of volatilities. The volatilities of all variable, in general, are the lowest under floating exchange rates. Volatilities under different kind of managed regimes lie between volatilities under fixed and floating exchange rate regimes.

The stabilization analysis gives suggestions on the undesirability of fixing or even managing exchange rate for the home country. A more stringent result on the desiderability of floating exchange rate as stabilization tools comes from the welfare evaluation.

I will adopt a measure of the welfare costs of business cycles given by the fraction of non-steady state consumption that households would be willing to give up in order to be indifferent between a constant sequence of consumption and working hours and the stochastic sequences of the same variables under the monetary regime considered. Appendix 9.5 shows

$$E\{X_t\} = AX_{t-1} + b\varepsilon_t$$

where  $X_t$  is the matrix of the endogenous variables at time t, A is the transition matrix and  $\varepsilon_t$  is the vector of the exogenous shocks which are assumed i.i.d. with unitary covariance. Let  $\Psi = b * \Sigma_{\varepsilon} * b'$  denote the variance covariance matrix of exogenous shocks. The matrix of the second moments  $\Omega$  of the endogenous variables is:

$$\Omega_{as} = \lim_{k \to \infty} \{ \sum_{i=0}^{k} (A^i) \Psi(A^i)' \}.$$

I calculated the second moments by approximating  $\Omega_{as}$  by  $\Omega_{k+1}$  so that the max[ $\Omega_{k+1} - \Omega_k$ ]  $\geq 1.0e - 0.8$ , where max stands for the maximum distance between any two elements of the matrix  $\Omega_n - \Omega$ .

<sup>37</sup>For this experiment I calibrated the persistence of the shock as 0.9, that is a reasonable value for any persistent shock. The variance of the shock is the same for the two economies and normalized so that the second moments of the variables result to be less than one. Values of the volatilities that range between zero and one will be useful to deliver positive number for the welfare comparison and will help to give a more clear interpretation of the welfare ranking. Different values for this variance will generate the same relative ranking among volatilities even with higher absolute values.

 $<sup>^{36}</sup>$  Volatility was computed using the following approximation procedure over the matrix of the second moments. Lets define the reduced form of the loglinearized model as follows:

the derivations of the welfare measure that assumes the following form:

$$v = 1 - \left[\frac{1}{2}((1-\sigma)E(\stackrel{\wedge}{c_t})^2 + (1+\phi)E(\stackrel{\wedge}{n_t})^2\right]$$
(35)

Given the parametrization of the calibrated model, where  $\sigma$  has been set equal 1 in order to get steady state balanced growth, the welfare measure is mainly determined from the volatility of labor. Table 3 shows the welfare gains for home and foreign technology shocks under different regimes in presence of financial frictions and interest rate smoothing. One can clearly see that the stabilization properties of floating exchange rates deliver the highest gain (and the lowest welfare cost that is the specular image), while fixed exchange rate regimes imply the higher welfare cost. The welfare gains from managed exchange rates lie between the two even tough they are much closer to the ones under floating exchange rates<sup>38</sup>. In addition one can notice that managing the change in the exchange rate improves welfare compared to management of the level as a result of the benefic effect on consumption of a higher stabilization in the price of the international bond<sup>39</sup>.

Tables 4 and 5 provide a systematic assessment of the welfare costs associated to fixed exchange rate regimes under different degrees of financial frictions and interest rate smoothing for both, home and foreign technology shocks. The welfare gains strongly decrease when we move to the model with endogenous capital accumulation and to the model with both, adjustment costs and agency costs. Higher degrees of interest rate smoothing tend to increase the welfare gains (and decrease welfare costs that are a specular image) across models and induce a very similar welfare gains in the models with and without agency costs. Finally table 6 shows that the difference in the welfare costs between fixed and floating exchange rates decreases when there are no agency costs or when an higher interest rate smoothing is applied. Notice that with an high interest rate smoothing the welfare differential between the two exchange rate regimes is almost the same both, in the model with agency costs and in the model without agency costs. This proves that when the monetary authority smooths the interest rate the impact of financial frictions is almost insignificant.

<sup>&</sup>lt;sup>38</sup>From the stabilization analysis they appeared much more destabilizing because the stabilization analysis looked at output and inflation volatility, while the welfare measure looks only at employment volatility.

<sup>&</sup>lt;sup>39</sup>Management in the change in the exchange rate helps the monetary authority to get rid of non stationary elements in the exchange rates and consequently in the price of the international bond.

We can then conclude that in presence of financial frictions fixed exchange rates deliver the highest welfare cost and that this cost tend to decrease when we shut off financial frictions or when we increase the degree of interest rate smoothing.

### 8 Conclusions

The paper addresses the issue of policy stabilization in terms of optimal choice of exchange rates regimes and the transmission mechanism of shocks in a two country model where financial frictions are explicitly microfounded. Financial stability can be a concern for the policy makers as long as the presence of financial frictions can impact the overall stability of the economy. In choosing the monetary regime a central bank of an open economy should take into account the impact of its choice on the real economy facing financial frictions. In presence of adjustment costs and agency costs, movements in real interest rates due to management of the exchange rates have a large impact on investment and net worth that are an important determinant of the business cycle. The insulation property of the mundellian theory holds also in a model without capital even though a model characterized by adjustment costs, that determine an endogenous variation in capital stock, and agency costs, that generate amplification in the cycle of the economy, enhances the result, provides a much richer explanation of the transmission mechanism behind this result and accounts for different sign in the transmission of external shocks. A clear ranking is identified among different monetary regimes in terms of persistence in the patterns of the impulse response, of volatilities of variables and welfare costs of the business cycle. In addition both, stabilization and welfare analysis, show that the presence of financial frictions destabilizes the economy and induces an extra burden mostly under fixed exchange rate regimes. The paper shows that interest rate smoothing can be a key determinant in stabilizing the economy. When the monetary authority adopts higher interest rate smoothing the dynamic properties of the economy with capital and financial frictions approach to those of the economy without capital and the welfare costs tend to decrease.

The paper seems to imply that if one country unilaterally adjusts its nominal interest rate in order to manage or fix exchange rates while the other country follows a Taylor rule, there can be large destabilizing effects when the economy is hit by foreign shocks. This suggests that large currency areas that are relatively closed should take advantage of the insulation properties of exchange rate regimes.

### References

- [1] Allen, F. and D. Gale (2000). "Bubble and Crisis", Economic Journal, 110, 236-256.
- [2] Anderson, G. and G. Moore (1985). "A Linear Procedure for Solving Linear Perfect Foresight Models", *Economics Letters*, 17, 247-252.
- [3] Aghion, P., P. Bacchetta and A. Banerjee (2000). "A Simple Model of Monetary Policy and Currency Crisis", *European Economic Review*, 44, 728-738.
- [4] Akerlof, G. (1970). "The Market for Lemons: Quality Uncertainty and The Market Mechanism", *Quarterly Journal of Economics*, 85, August, 488-500.
- [5] Ball, L. (1998). "Policy Rules for Open Economies", in J. Taylor, ed., Monetary Policy Rules, University of Chicago Press, Chicago, IL.
- [6] Benigno, G., P. Benigno and F. Ghironi (2000). "Interest Rate Rules for Fixed Exchange Rate Regimes", manuscript, Bank of England, New York University and Boston College.
- [7] Bernanke, B. and M. Gertler (1989) "Agency Costs, Net Worth, and Business Fluctuations", American Economic Review, 79, March 1989, 14-31.
- [8] Bernanke, B., M. Gertler and S. Gilchrist (1999). "The Financial Accelerator in a Quantitative Business Cycle Framework", in J.B. Taylor, and M. Woodford, eds., *Handbook* of Macroeconomics, Amsterdam: North-Holland.
- [9] Blanchard, O. and M. Watson (1982). "Bubbles, Rational Expectations and Financial Markets", in P. Watchel, ed., Crisis in the Economic and Financial Structure, Lexington Books.
- [10] Burniside C., M. Eichenbaum and S. Rebelo (2000). "Hedging and Financial Fragility in fixed Exchange Rate Regimes", manuscript, Northwestern University, October.
- [11] Calvo, G. (1983). "Staggered Prices in a Utility Maximization Framework", Journal of Monetary Economics, 12, 383-98.
- [12] Carlstrom, C. and T. Fuerst (1997). "Agency Costs, Net Worth and business Fluctuations: A Computable General Equilibrium Analysis", *American Economic Review*, 87, 893-910.
- [13] Chang, R., Cespedes, L. and A. Velasco (2000). "Balance Sheets and Exchange Rate Policy", NBER Working Paper No. 7840, August.
- [14] Clarida, R., J. Galí, and M. Gertler (1998). "Monetary Policy Rules in Practice: Some International Evidence", *European Economic Review*, 42 (6), June, 1033-1067.
- [15] Clarida, R., J. Gali, and M. Gertler (2000). "Monetary Policy Rules and Macroeconomic Stability: Evidence and Some Theory", *Quarterly Journal of Economics*, 115 (1), February, 147-180.
- [16] Cooley, T. and V. Quadrini (1999). "Monetary Policy and the Financial Decision of Firms", manuscript, New York University.
- [17] Devereux, M. and C. Engel (2000) "Monetary Policy in the Open Economy Revisited: Price Setting and Exchange Rate Flexibility", NBER Working Paper No. 7655.
- [18] Gale, D. and M. Hellwig (1985). "Incentive-Compatible Debt Contracts: The One-Period Problem", *Review of Economic Studies*, 52, 647-663.
- [19] Galí, J. and T. Monacelli (2000). "Optimal Monetary Policy and Exchange Rate Volatility in A Small Open Economy", manuscript, Universitat Pompeu Fabra and Boston College.
- [20] M. Gertler, Gilchrist S. and F.M. Natalucci (2001) "External Constraints on Monetary Policy and The Financial Accelerator", mimeo, New York University and Boston University.
- [21] Kiyotaki, N. and J. Moore (1997). "Credit Cycles", Journal of Political Economy, 105, April, 211-48.

- [22] McCallum, B. and E. Nelson (1999). "Nominal Income Targeting in an Open-Economy Optimizing Model", Journal of Monetary Economics, 43: 533-578.
- [23] Monacelli, T. (1999). "Into the Mussa Puzzle: Monetary Policy Regimes and the Real Exchange Rate in a Small Open Economy", manuscript, Boston College.
- [24] Obstfeld, M. and K. Rogoff (1995). "The Mirage of Fixed Exchange Rates", Journal of Economic Perspective, 9, 73-96.
- [25] Rotemberg, J. and M. Woodford (1997). "An Optimizing Based Econometric Framework for the Evaluation of Monetary Policy", in B. Bernanke and J. Rotemberg, eds., NBER Macroeconomics Annual, Cambridge, MA, MIT Press.
- [26] Svensson, L. (1999). "Inflation Forecast Targeting as a Monetary Policy Rule", Journal of International Economics, 50: 155-183.
- [27] Taylor, J. (1999). Monetary Policy Rules, University of Chicago Press, Chicago, IL.
- [28] Weeparana, A. (1998). "The Performance of Simple Monetary Policy Rules in a Large Open Economy", manuscript, Stanford University.
- [29] Woodford, M. (1999). "Inflation Stabilization and Welfare", manuscript, Princeton University.

# 9 Appendix

#### 9.1 Derivation of the Complete Loglinearized Model

#### 9.1.1 Aggregate Demand Equations

The households solve a three stage problem. I show the result for the home country: a similar solution holds for the foreign country.

First stage. To get the optimal quantity  $C_{s,t}$ , with s = H, F, we solve:

$$\max(C_{s,t} = \left(\int_0^1 C_{s,t}(i)^{\frac{\epsilon-1}{\epsilon}} di\right)^{\frac{\epsilon}{\epsilon-1}})$$
(36)

$$s.t.P_{s,t}C_{s,t} \le E_{s,t} \tag{37}$$

where  $C_{s,t}$  is a CES aggregator and  $E_{s,t}$  is the amount of expenditure allocated to the good  $C_{s,t}$ . The solutions of the two maximization problems are:

$$C_{s,t}(i) = \left(\frac{P_{s,t}(i)}{P_t}\right)^{-\varepsilon} C_t \tag{38}$$

where  $P_{s,t} = \left(\int_0^1 P_{s,t}(i)^{1-\epsilon} di\right)^{\frac{1}{\epsilon-1}}$  is defined as the price that minimize the expenditure given the optimal quantity of consumption. Since there is no international price discrimination  $P_{F,t}(i) = e_t P_{H,t}^*(i), \forall i \in [0, 1]$ , where  $e_t$  is the nominal exchange rate expressed as the price of foreign currency in terms of the home currency and  $P_{H,t}^*(i)$  is the price of foreign good *i* denominated in foreign currency.

Second Stage. The consumers choose the optimal allocation between home and foreign good. They solve the following problem: choose  $C_{H,t}$  and  $C_{F,t}$  to:

$$MaxC_{t} \equiv \left[ (1-\gamma)^{1/\eta} C_{H,t}^{\frac{\eta-1}{\eta}} + \gamma^{\frac{1}{\eta}} C_{F,t}^{\frac{\eta-1}{\eta}} \right]^{\frac{\eta}{\eta-1}}$$
(39)

$$s.t.P_tC_t \le E_t \tag{40}$$

where  $\gamma$  is the share of foreign demand in the consumption index or the degree of openness of the home economy. The solution to this problem gives the optimal demand for home good from the home country and the foreign country:

$$C_{H,t} = (1-\gamma)(\frac{P_{H,t}}{P_t})^{-\eta}C_t; C_{F,t} = \gamma(\frac{P_{F,t}}{P_t})^{-\eta}C_t.$$
(41)

Similarly  $P_t \equiv [(1 - \gamma)P_{H,t}^{1-\eta} + \gamma P_{F,t}^{1-\eta}]^{\frac{1}{1-\eta}}$  is defined as the price that minimizes expenditure given the optimal allocation of consumption. The maximization problem for the foreign country is just the same and the degree of openness of the foreign country is denoted by  $\gamma^*$ . I assume  $\gamma^* = \gamma$ . The loglinearized expressions for the optimal allocations of consumption between home and foreign goods for the home economy are:

$${}^{\wedge H}_{c_t} = -\eta ({}^{\wedge H}_t - {}^{\wedge}_{p_t}) + {}^{\wedge}_{c}; {}^{\wedge F}_{c_t} = -\eta ({}^{\wedge F}_t - {}^{\wedge}_{p_t}) + {}^{\wedge}_{c_t}$$
(42)

and for the foreign economy:

$$\hat{c}_{t}^{H*} = -\eta(\hat{p}_{t}^{H*} - \hat{p}_{t}^{*}) + \hat{c}_{t}^{*}; \ \hat{c}_{t}^{F*} = -\eta(\hat{p}_{t}^{F*} - \hat{p}_{t}^{*}) + \hat{c}_{t}^{*}.$$
(43)

Let us now define the terms of trade as:  $s_t = \frac{P_{F,t}}{P_{H,t}} = \frac{e_t P_{F,t}}{P_{H,t}}$ . The loglinearized expression for the terms of trade is:  $\hat{s}_t = \hat{p}_t^F - \hat{p}_t^H = \hat{e}_t + \hat{p}_t^{\wedge F} - \hat{p}_t^H$ . Combining this expression with the loglinearized expression for the consumption index for the home country  $\hat{p}_t = [(1-\gamma)\hat{p}^F + \gamma \hat{p}^H]$  and rearranging the consumption allocation as function os the terms of trade I get:

$$\overset{\wedge}{c}_{t}^{H} = \eta (1-\gamma) \overset{\wedge}{s}_{t} + \overset{\wedge}{c}_{t}; \ \overset{\wedge}{c}_{t}^{F} = -\eta \gamma \overset{\wedge}{s}_{t} + \overset{\wedge}{c}_{t}.$$

$$(44)$$

Assuming  $\stackrel{\wedge}{s_t} = -\stackrel{\wedge^*}{s_t}$  for the foreign country I get:

$$\hat{c}_{t}^{*H} = \eta (1 - \gamma^{*}) \hat{s}_{t} + \hat{c}_{t}^{*}; \ \hat{c}_{t}^{*F} = -\eta \gamma^{*} \hat{s}_{t} + \hat{c}_{t}^{*}.$$
(45)

Third Stage. Households choose total consumption, the quantity of labor and they make portfolio decisions in bank deposits and bonds. The result of this stage of the maximization are the FOC (4), (5). Assuming a utility of the form  $U_t = \frac{C_t^{1-\sigma}}{1-\sigma} - \frac{N_t^{1-\phi}}{1-\phi}$  and log-linearizing equations (4) and (8)we obtain the loglinear expressions for the home and foreign consumption demands:

$$\widehat{c}_{t}^{-\sigma} = E_{t}\{\widehat{c}_{t+1}\}^{-\sigma} - \frac{1}{\sigma}(\widehat{r}_{t}^{n} - E_{t}\{\widehat{\pi}_{t+1}\}); \widehat{c}_{t}^{*-\sigma} = E_{t}\{\widehat{c}_{t+1}^{*}\} - \frac{1}{\sigma}(\widehat{r}_{t}^{n*} - E_{t}\{\widehat{\pi}_{t+1}^{*}\})$$
(46)

If we add a shock to preferences of the form  $\hat{\nu}_t = \rho_{\nu}\hat{\nu}_{t-1} + \varepsilon_t^{\nu}$  (that enters the utility function in the following way:  $E_t \sum_{k=0}^{\infty} \beta^k [\nu_t(U(C_{s,t+k}(C^H, C^F)) - V_s(N_{t+k})])$  we get the following loglinear expressions for the consumption demand:

$$\widehat{c}_{t}^{-\sigma} = E_{t}\{\widehat{c}_{t+1}\}^{-\sigma} - \frac{1}{\sigma}(\widehat{r}_{t}^{n} - E_{t}\{\widehat{\pi}_{t+1}\}) - \frac{1}{\sigma}(\nu_{t+1} - \nu_{t});$$
(47)

$$\widehat{c}_{t}^{*-\sigma} = E_{t}\{\widehat{c}_{t+1}^{*}\} - \frac{1}{\sigma}(\widehat{r}_{t}^{n*} - E_{t}\{\widehat{\pi}_{t+1}^{*}\}) - \frac{1}{\sigma}(\nu_{t+1}^{*} - \nu_{t}^{*})$$

$$(48)$$

Let us now look at the loglinearized expression for the UIP. By loglinearizing (7)(under the assumption of certainty equivalence) a standard form of the uncovered interest parity holds:

$$\widehat{r}_t^n - \widehat{r}_t^{n*} = E_t \{ \Delta e_{t+1} \}$$

$$\tag{49}$$

where  $\hat{r}_t^n = \log(\frac{R_t^n}{R^n})$  and  $\hat{r}_t^{n*} = \log(\frac{R_t^{n*}}{R^{n*}})$ . By using the terms of trade equation in log deviations and first differencing equation (49) and combining the two expressions I get:

$$\widehat{s}_t = (\widehat{r}_t^{n*} - E_t\{\widehat{\pi}_{F,t+1}^*\}) - (\widehat{r}_t^n - E_t\{\widehat{\pi}_{H,t+1}\}) + E_t\{\widehat{s}_{t+1}\}.$$
(50)

In addition one can show that:

$$\pi_t = \pi_{H,t} + \gamma \Delta \hat{s}_t; \ \pi_t^* = \pi_{Ft}^* - \gamma^* \Delta \hat{s}_t.$$
(51)

Defining the real exchange rate as  $e_t^R = \frac{e_t P_t^*}{P_t}$  the following relation between the real exchange rate and the terms of trade holds:  $e_t^R = (1 - 2\gamma) \hat{s}_t$ . By substituting the real exchange rate in the (51) and imposing  $\gamma = \gamma^*$  we get:

$$\pi_t = \pi_{H,t} + \frac{\gamma}{1 - 2\gamma} \Delta \hat{s}_t; \ \pi_t^* = \pi_{F,t} + \frac{\gamma}{1 - 2\gamma} \Delta \hat{s}_t.$$
(52)

By substituting the last relation for the home and foreign country in the UIP we can get an expression of the UIP in real terms. Notice that if we assume the presence of some credibility shocks (shocks to the terms of trade) the UIP can be written in the following way:

$$\widehat{s}_t = (\widehat{r}_t^{n*} - E_t\{\widehat{\pi}_{F,t+1}^*\}) - (\widehat{r}_t^n - E_t\{\widehat{\pi}_{H,t+1}\}) + E_t\{\widehat{s}_{t+1}\} + (f_t).$$
(53)

where  $f_t = \rho_f f_{t-1} + \varepsilon_f$ . Finally we need to derive a loglinear expression for the aggregate demand of both countries. Market clearing in the domestic and the foreign economy, under a symmetric equilibrium, implies:

$$Y_{H,t} = C_{H,t} + C_{H,t}^* + I_{H,t}; \ Y_{F,t} = C_{F,t} + C_{F,t}^* + I_{F,t}$$
(54)

where  $I_t$  is aggregate investment and  $C_t^*$  is foreign demand. Log-linearization around a steady state with balanced trade implies:

$$\widehat{y}_{t} = \zeta_{c_{h}}\widehat{c}_{H,t} + \zeta_{c_{h^{*}}}\widehat{c}_{H,t}^{*} + \zeta_{I_{h}}\widehat{i}_{H,t}; \ \widehat{y}_{t} = \zeta_{c_{f}}\widehat{c}_{F,t} + \zeta_{c_{f^{*}}}\widehat{c}_{F,t}^{*} + \zeta_{I}\widehat{i}_{F,t}$$
(55)

where  $\zeta_j$  is the share of variable j in total domestic output<sup>40</sup>.

#### 9.2 The Calvo Producers

The monopolistic sector producers choose  $P_{H,t}(i)$  to:

$$Max \sum_{k=0}^{\infty} (\vartheta)^{k} E_{t} \left\{ \Lambda_{t,t+k} [Y_{t+k}(i)(P_{H,t}(i) - \mu MC_{t})] \right\}$$
(56)

s.t.

$$Y_{t+k}(i) = \left(\frac{P_{H,t}(i)}{P_{H,t+k}}\right)^{-\varepsilon} (C_{H,t+k} + C_{H,t+k}^*)$$
(57)

where  $\mu$  is a mark-up,  $\vartheta$  is the probability that the price is fixed in each period,  $C_{H,t}(i) = (\frac{P_{H,t}(i)}{P_{H,t}})^{-\varepsilon}C_{H,t}^*$ ;  $C_{H,t}^*(i) = (\frac{P_{H,t}^*(ij)}{P_{H,t}^*})^{-\varepsilon}C_{H,t}^*$ , and  $\Lambda_{t,t+k} = \beta^k(\frac{C_t}{C_{t+k}})(\frac{P_t}{P_{t+k}})$  is the discount rate and  $MC_t^{-41}$  is the nominal marginal cost  $(\frac{W_t}{MC_t} = (1-\alpha)\frac{Y_{H,t}(k)}{N_t(k)})$ .

Maximization yields the following first order condition in terms of stationary variables:

$$\sum_{k=0}^{\infty} (\vartheta\beta)^k E_t \left\{ (C_{t+k})^{-\sigma} Y_{t+k}(i) [(P_{H,t}(i)/P_{H,t} - \mu(\frac{P_{t+k}}{P_t})mc_t)] \right\} = 0.$$
(58)

Since  $(1 - \vartheta)$  firms change their prices the price index can be written as:  $P_{H,t} \equiv [\vartheta P_{H,t-1}^{1-\epsilon}(i) + (1 - \vartheta)P_{H,t}^{1-\epsilon}(i)]^{\frac{1}{1-\epsilon}}$ . Loglinearizing the price index and the condition (58) around a perfect

$$\begin{split} \zeta_h &= (1-\gamma) [1 - \frac{\delta \alpha}{\mu (R-1+\delta)}] \\ \zeta_{h^*} &= \frac{\gamma}{1-\gamma} \zeta_h \\ \zeta_I &= \frac{\delta \alpha}{\mu (R-1+\delta)} \end{split}$$

 $^{41}$ The marginal cost for the monopolistic sector is actually equal to the unit price of the homogenous good used as input. Although the homogenous good is produced by a competitive sector that sets the price of the homogenous good equal to the marginal cost of labor.

 $<sup>^{40}</sup>$ These coefficients are function of the degree of openness and the steady state ratio of the variables. In particular:

foresights, zero inflation, steady state trade balance, after some algebra, we obtain the following aggregate supply equation:

$$\widehat{\pi}_{H,t} = \beta E_t \{ \widehat{\pi}_{H,t+1} \} + \lambda \widehat{mc}_t \tag{59}$$

where  $\lambda \equiv \frac{(1-\vartheta)(1-\beta\vartheta)}{\vartheta}$ .

### 9.3 The Financial Intermediary

The agency problem is solved by assuming that the intermediary chooses the quantities of loans and the repayment schedule (or alternatively the capital demand) so as to maximize the expected return of the risk neutral entrepreneur subject to a participation constraint for the risk neutral intermediary<sup>42</sup> and a participation constraint for the borrower<sup>43</sup>. Firm j chooses  $K_{t+1}^{j}$ ,  $\overline{\omega}^{j}$  to

$$MaxE\{\int_{\overline{\omega}}^{\infty} (\omega - \overline{\omega}^{j}) R_{t+1}^{k,j} Q_t K_{t+1}^j dF(\omega)\}$$
(60)

$$\{[1 - F(\bar{\omega}^{j})](R_{L,t+1}^{j}L_{t+1}^{j}) + (1 - m)\int_{0}^{\bar{\omega}^{j}}\omega dF(\omega)\}R_{t+1}^{k,j}Q_{t}K_{t+1}^{j}$$
(61)

$$= R_{t+1}D_{t+1}^{j}(\frac{P_{t}}{P_{H,t}})$$

$$\bar{\omega}^{j}R_{t+1}^{k}Q_{t}K_{t+1}^{j} = R_{L,t+1}^{j}L_{t+1}^{j}$$
(62)

 $<sup>^{42}</sup>$ We can look at this participation contraints from two different perspectives. On one side it says that the expected return that the risk neutral intermediary gets from the loan activity has to be equal to the expected return that he would get from a risk free portfolio: in equilibrium the firm extracts all the surplus and the intermediary is just indifferent. On the other side, the intermediary acts as a competitive agent so we need to impose a zero-profit condition stating that the expected return from the investment in risky projects has to be equal to the return that the intermediary has to pay on deposits: the intermediary is just an interface between the investors and firms and its existence is justified by the fact that by pooling different investment projects the intermediary is able to get an expected return that in the limit, for n that approaches  $\infty$ , becomes known.

 $<sup>^{43}</sup>$ The borrower acts in a competitive manner so its return on capital has to equate the total payments from loans. Looking at it from another perspective we can say that the firm has to be indifferent between investing in capital goods and investing in a balanced portfolio.

$$(Q_t K_{t+1}^j - \Pi_{t+1}^j) = L_{t+1}^j \tag{63}$$

where  $\omega^j$  is the idiosyncratic shock faced by the entrepreneur and distributed with function  $F(\omega)$ ,  $\omega^{j}$  is value of the shock that divides the random space into default and solvency regions,  $R_{L,t+1}^j$  and  $R_{L,t+1}^j$  are the repayment schedules required for loans denominated in domestic and foreign currency,  $L_t$  and  $L_t^*$  are the amount of loan denominated in domestic currency and foreign currency, (m) is the monitoring cost paid by the lender,  $\Pi_{t+1}^j$  is the net worth of the entrepreneur, finally  $\xi$  is the fraction of loans denominated in foreign currency, finally  $R_{t+1}^k(j) = (\frac{mc_t(j)\gamma \frac{\alpha A_t K(j) \Gamma^2}{K_t(j)} + Q_{t+1}(1-\delta)}{Q_t})$  where  $A_t$  is subject to an aggregate shock. Equation (60) is the expected return to the entrepreneur, equation (61) is the participation constraint of the lender, equation (62) is the participation constraint for the borrower, finally equation (63) and (??) define the amount of loans in both domestic and foreign currency. All quantities are expressed in term of home consumption goods.

Lets define  $\int_{\omega}^{\infty} (\omega - \overline{\omega}^{j}) = (1 - F(\overline{\omega}^{j})) \equiv (1 - \Gamma(\overline{\omega}^{j})), \widetilde{a}$  as the random variable for the aggregate shock to technology  $A_t$ ,  $\Phi = \frac{R^k}{R(\frac{P}{P_H})}$  as the ratio between the return on capital and the cost of loan and  $k = \frac{QK}{W}$  as the ratio between the value of capital and the net worth. Given this and using the equilibrium conditions for loans and deposits, we can rewrite the maximization problem for each borrower j in the following way:

$$Max_{k,\tilde{\omega}}^{j}E(1-\Gamma(\tilde{\omega}^{j}))(\Phi k\tilde{a})$$
(64)

$$s.to[\Gamma(\tilde{\omega}^{j}) - mF(\tilde{\omega}^{j})](\Phi k\tilde{a}) = [k-1]$$
(65)

The first order conditions with respect to  $k, \overline{\omega}^{j}$  and the Lagrange multiplier  $\lambda$  are:

$$E\{([(1 - \Gamma(\widetilde{\omega}^{j})] + \lambda(\Gamma(\widetilde{\omega}^{j}) - mF(\widetilde{\omega}^{j}))] * (\Phi\widetilde{a})) - \lambda\} = 0$$
(66)

$$\Gamma'(\overline{\omega}^{j}) - \lambda[\Gamma'(\overline{\omega}^{j}) - mF'(\overline{\omega}^{j})] = 0$$
(67)

$$[\Gamma(\overline{\omega}^{j}) - mF(\overline{\omega}^{j})](\Phi k\widetilde{a}) - [k-1] = 0$$
(68)

Solving the first order conditions, one can prove that a linear inverse relation exists between the net worth/ capital ratio (k) and the risk premium adjusted to take into account the normalization effect due to the terms of trade  $(\Phi)$ .

### 9.4 The Steady State of the Economy

Let us characterize the perfect foresight steady state of the two country world economy, taking Y (where  $Y = Y_H = Y_F$  in steady state), the output level in the flexible price case for both countries, as given and A = 1. I will use variables without time subscripts to refer to steady state values. Markups are constant in the steady state, implying a product wage  $MC = \frac{1}{\mu}$ . From the Euler in steady state we get  $R = \frac{1}{\beta}$ . Given that Q = 1 and  $MPK = mc * \alpha \frac{Y}{K} = \frac{1}{\mu} * \alpha \frac{Y}{K}$ , the return on capital in steady state is  $R^k = \frac{1}{\mu} \alpha \frac{Y}{K} + (1 - \delta) = R + \varkappa$ , where  $\varkappa$  is the risk premium in steady state. From that I get  $\frac{Y}{K} = \frac{\mu(R^{k}-1+\delta)}{\alpha}$ . The law for capital accumulation in the steady state holds as  $K = K(1 - \delta) + \phi(\frac{1}{K})k$  and  $\frac{1}{K} = \delta$  in the steady state. Using the last ratio we get that:  $\frac{1}{Y} = \frac{\delta \alpha}{\mu(R^k-1+\delta)}$ . Consider a steady state where initial costs are normalized so that  $e^R = 1$  and s = 1. This implies that in a balance growth path trade balance are equal to zero or that  $C_F = C_{H^*}$ . Given this assumption the following equality holds:  $\frac{C_H}{Y} = \frac{\gamma}{1-\gamma} \frac{C_F}{Y} = \frac{\gamma}{1-\gamma} \frac{C_{H^*}}{Y}$ . In the loglinearized version of the resource constraint in steady state we find that in steady state the following ratios hold:  $\frac{C_H}{Y} = \gamma [1 - \frac{\delta \alpha}{\mu(R^k-1+\delta)}]; \frac{C_F}{Y} = (1 - \gamma)[1 - \frac{\delta \alpha}{\mu(R^k-1+\delta)}] = \frac{C_{H^*}}{Y}$ . In the loglinearized version of the resource constraint  $\zeta_h = \frac{C_H}{Y}, \zeta_{h^*} = \frac{C_{H^*}}{Y}, \zeta_{I_h} = \frac{K}{Y}$ .

### 9.5 The Welfare Measure

If  $C_t$  and  $N_t$  are the equilibrium stochastic processes of consumption and labor corresponding to a particular monetary policy, the cost of business cycles under such policy will be measured by v that satisfies the certainty equivalence relation:

$$U((1-v)C_t) - V(N_t) = E_t \{ U(C_t) - V(N_t) \}$$
(69)

where  $E_t$  is the mathematical expectation. The business cycle associated with a particular monetary policy will be costly if v is positive. Let's assume that consumption and labor are distributed as Gaussians. From the first order approximation to the equation (69), the measure v can be approximated by:

$$v \approx \frac{E_t(U(C_t, N_t)) - U(C_{ss}, N_{ss})}{U_{\wedge}(\hat{c}_{ss}, \hat{n}_{ss})}$$
(70)

where  $E_t(U(C_t, N_t))$  is the expected utility,  $U(C_{ss}, N_{ss})$  is the utility evaluated at the steady state and  $U_{\hat{c}}(\hat{c}_{ss}, \hat{n}_{ss})$  is the first derivative of the utility with respect to the logarithm of  $C_t$  around the logarithm of C. Approximating the expected utility with a second order Taylor expansion with respect to the logarithm of  $C_t$  and  $N_t$  around the logarithm of C and N and substituting the expression in the approximation for v we get the welfare measure. The Hessian of the utility function with respect to the log deviations of consumption and labor from the steady state is:

$$H = \begin{pmatrix} C_{ss}(1-\sigma)\exp(1-\sigma)\hat{c}_t & 0\\ 0 & N_{ss}(1+\phi)\exp(1+\phi)\hat{n}_t \end{pmatrix}.$$
 (71)

Assuming that  $\log(\frac{y_y}{y_{ss}}) = 0$  where y = C, N, then we can write the second order Taylor expansion for the expected utility as:

$$E_t(U(C_t, N_t)) \approx U(C_{ss}, N_{ss}) + \frac{1}{2}C_{ss}(1-\sigma)\exp(1-\sigma)\hat{c}_t E(\hat{c}_t)^2$$
 (72)

$$+\frac{1}{2}N_{ss}(1+\phi)\exp(1+\phi)\hat{n}_{t}E(\hat{n}_{t})^{2}+''O''$$
(73)

where  $E(\hat{c}_t)^2$  and  $E(\hat{n}_t)^2$  are the second moments of consumption and labor. Assuming the following utility function  $U_t = \frac{C_t^{1-\sigma}}{1-\sigma} - \frac{N_t^{1+\phi}}{1+\phi}$  and substituting (72) in (70) we get:

$$\upsilon = 1 - \left[\frac{1}{2}((1-\sigma)E(\hat{c}_t)^2 + (1+\phi)E(\hat{n}_t)^2\right].$$
(74)

## Table I

### Home Technoogy Shock: Volatility (with financial frictions)\*

Exchange Rate Regime	Floating Fixed	Managed
----------------------	----------------	---------

Variables			$e_t - \bar{e}^{**}$	$e_t - e_{t-1} ***$
Output	0.06	0.35	0.046	0.064
CPI Inflation	0.012	0.27	0.033	0.011
Consumption	0.09	0.33	0.12	0.07
Employment	0.020	0.59	0.044	0.016
Capital	0.008	0.092	0.005	0.025
Investment	0.018	0.66	0.02	0.038
Price of Capital	0.03	1.33	0.04	0.057
Interest Rate	0.006	0.11	0.008	0.008
Return on Capital	0.021	0.73	0.015	0.024
Terms of Trade	0.038	1.19	0.12	0.011

\* The calculations have been made assuming a degree of openness of 0.15 and assuming that the foreign country follows a Taylor rule symmetric to the one of the home country.

\*\*This column evaluates volatility under a Taylor rule that reacts

to the deviations of the exchange rate from the steady state.

\*\*\* This column evaluates volatility under a Taylor rule that reacts

to the change in the exchange rates from t-1 and t.

## Table 2

Foreign Technology Shock: Volatility (with financial frictions)\*

Exchange frate fregulie	rioaung	I IACU	Manageu	
Variables			$e_t - \bar{e}^{**}$	$e_t - e_{t-1} ***$
Output	0.011	0.67	0.033	0.008
CPI Inflation	0.0049	0.46	0.034	0.004
Consumption	0.026	0.47	0.05	0.016
Employment	0.015	0.97	0.045	0.012
Capital	0.0085	0.14	0.012	0.004
Investment	0.019	1.10	0.05	0.01
Price of Capital	0.036	2.22	0.1	0.019
Interest Rate	0.0016	0.20	0.005	0.01
Return on Capital	0.011	1.22	0.043	0.018
Terms of Trade	0.05	1.94	0.14	0.020

 $^{\ast}$  The calculations have been made assuming a degree of openness of 0.15 and assuming that

the foreign country follows a Taylor rule symmetric to the one of the home country.

 $^{\ast\ast}$  This column evaluates volatility under a Taylor rule that reacts

to the deviations of the exchange rate from the steady state.

 $^{\ast\ast\ast}$  This column evaluates volatility under a Taylor rule that reacts

to the change in the exchange rates from t-1 and t.

# Table 3

Welfare gain comparison of different regimes in the model with agency costs and interest rate smoothing of 0

Welfare	Home Technology Shock	Foreign Technology Shock
Floating Exchange Rates	0.98	0.984
Fixed Exchange Rates	0.41	0.03
Managed on $e_t$	0.953	0.95
Managed on $\Delta e_t$	0.989	0.98

### Table 4

Home Technology Shock. Comparison of welfare gains under fixed exchange rates for different degrees of financial frictions and interest rate smoothing

Welfare	Model Without Capital	No Agency Costs	With Agency Costs
$\chi = 0$	0.978	0.51	0.41
$\chi = 0.9$	0.993	0.84	0.84

### Table 5

Foreign Technology Shock. Comparison of welfare gains under fixed exchange rates for different degrees of financial frictions and interest rate smoothing

Welfare	Model Without Capital	No Agency Costs	With Agency Costs
$\chi = 0$	0.9736	0.47	0.3
$\chi = 0.9$	0.973	0.849	0.842

### Table 6

Welfare differential between fixed and floating under both technology shocks across models

Welfare Differential: $v(\text{fix}) - v(\text{float})$	Home Techn Shock	Foreign Techn Shock
Agency Costs, $\chi = 0$	0.57	0.954
No Agency Costs, $\chi = 0$	0.465	0.514
Agency Costs, $\chi = 0.9$	0.136	0.1517
No Agency Costs, $\chi = 0.9$	0.136	0.1445

# Figure I

Impulse response to home technology shock in the model with capital and agency costs: floating (solid line) versus fixed (dashed line)



Impulse response to home technology shock in the model without capital: floating (solid line) versus fixed (dashed line) exchange rates



Impulse response to home technology shock in the model with capital, agency costs and interest rate smoothing: floating (solid line) versus fixed (dashed line) exchange rates



Impulse response to foreign technology shock in the model with capital and agency costs: floating (solid line) versus fixed (dashed line) exchange rates



Impulse response to foreign technology shock in the model with capital and no agency costs: floating (solid line) versus fixed (dashed line) exchange rates







Impulse response to foreign technology shock in the model with capital, agency costs and interest rate smoothing: floating (solid line) versus fixed (dashed line) exhange rates



Impulse response to home technology shock in the model with capital and agency costs: floating, versus managed on change, versus on level, versus fixed exchang rates



Impulse response to foreign technology shock in the model with capital and agency costs: floating, versus managed on change, versus on level, versus fixed exchange rates



## **European Central Bank Working Paper Series**

- I "A global hazard index for the world foreign exchange markets" by V. Brousseau and F. Scacciavillani, May 1999.
- 2 "What does the single monetary policy do? A SVAR benchmark for the European Central Bank" by C. Monticelli and O. Tristani, May 1999.
- 3 "Fiscal policy effectiveness and neutrality results in a non-Ricardian world" by C. Detken, May 1999.
- 4 "From the ERM to the euro: new evidence on economic and policy convergence among EU countries" by I.Angeloni and L. Dedola, May 1999.
- 5 "Core inflation: a review of some conceptual issues" by M.Wynne, May 1999.
- 6 "The demand for M3 in the euro area" by G. Coenen and J.-L. Vega, September 1999.
- 7 "A cross-country comparison of market structures in European banking" by O. de Bandt and E. P. Davis, September 1999.
- 8 "Inflation zone targeting" by A. Orphanides and V. Wieland, October 1999.
- 9 "Asymptotic confidence bands for the estimated autocovariance and autocorrelation functions of vector autoregressive models" by G. Coenen, January 2000.
- 10 "On the effectiveness of sterilized foreign exchange intervention" by R. Fatum, February 2000.
- II "Is the yield curve a useful information variable for the Eurosystem?" by J. M. Berk and P. van Bergeijk, February 2000.
- 12 "Indicator variables for optimal policy" by L. E. O. Svensson and M. Woodford, February 2000.
- 13 "Monetary policy with uncertain parameters" by U. Söderström, February 2000.
- 14 "Assessing nominal income rules for monetary policy with model and data uncertainty" by G. D. Rudebusch, February 2000.
- 15 "The quest for prosperity without inflation" by A. Orphanides, March 2000.
- 16 "Estimating the implied distribution of the future short term interest rate using the Longstaff-Schwartz model" by P. Hördahl, March 2000.
- 17 "Alternative measures of the NAIRU in the euro area: estimates and assessment" by S. Fabiani and R. Mestre, March 2000.
- 18 "House prices and the macroeconomy in Europe: Results from a structural VAR analysis" by M. lacoviello, April 2000.

- 19 "The euro and international capital markets" by C. Detken and P. Hartmann, April 2000.
- 20 "Convergence of fiscal policies in the euro area" by O. De Bandt and F. P. Mongelli, May 2000.
- 21 "Firm size and monetary policy transmission: evidence from German business survey data" by M. Ehrmann, May 2000.
- 22 "Regulating access to international large value payment systems" by C. Holthausen and T. Rønde, June 2000.
- 23 "Escaping Nash inflation" by In-Koo Cho and T. J. Sargent, June 2000.
- 24 "What horizon for price stability" by F. Smets, July 2000.
- 25 "Caution and conservatism in the making of monetary policy" by P. Schellekens, July 2000.
- 26 "Which kind of transparency? On the need for clarity in monetary policy-making" by B.Winkler, August 2000.
- 27 "This is what the US leading indicators lead" by M. Camacho and G. Perez-Quiros, August 2000.
- 28 "Learning, uncertainty and central bank activism in an economy with strategic interactions" by M. Ellison and N.Valla, August 2000.
- 29 "The sources of unemployment fluctuations: an empirical application to the Italian case" by S. Fabiani, A. Locarno, G. Oneto and P. Sestito, September 2000.
- 30 "A small estimated euro area model with rational expectations and nominal rigidities" by G. Coenen and V. Wieland, September 2000.
- 31 "The disappearing tax base: Is foreign direct investment eroding corporate income taxes?" by R. Gropp and K. Kostial, September 2000.
- 32 "Can indeterminacy explain the short-run non-neutrality of money?" by F. De Fiore, September 2000.
- 33 "The information content of M3 for future inflation" by C. Trecroci and J. L. Vega, October 2000.
- 34 "Capital market development, corporate governance and the credibility of exchange rate pegs" by O. Castrén and T. Takalo, October 2000.
- 35 "Systemic risk: A survey" by O. De Bandt and P. Hartmann, November 2000.
- 36 "Measuring core inflation in the euro area" by C. Morana, November 2000.
- 37 "Business fixed investment: Evidence of a financial accelerator in Europe" by P. Vermeulen, November 2000.

- 38 "The optimal inflation tax when taxes are costly to collect" by F. De Fiore, November 2000.
- 39 "A money demand system for euro area M3" by C. Brand and N. Cassola, November 2000.
- 40 "Financial structure and the interest rate channel of ECB monetary policy" by B. Mojon, November 2000.
- 41 "Why adopt transparency? The publication of central bank forecasts" by P. M. Geraats, January 2001.
- 42 "An area-wide model (AWM) for the euro area" by G. Fagan, J. Henry and R. Mestre, January 2001.
- 43 "Sources of economic renewal: from the traditional firm to the knowledge firm" by D. R. Palenzuela, February 2001.
- "The supply and demand for eurosystem deposits The first 18 months" by U. Bindseil and F. Seitz, February 2001.
- 45 "Testing the Rank of the Hankel matrix: a statistical approach" by G. Camba-Mendez and G. Kapetanios, February 2001.
- 46 "A two-factor model of the German term structure of interest rates" by N. Cassola and J. B. Luís, February 2001.
- 47 "Deposit insurance and moral hazard: does the counterfactual matter?" by R. Gropp and J.Vesala, February 2001.
- 48 "Financial market integration in Europe: on the effects of EMU on stock markets" by M. Fratzscher, March 2001.
- 49 "Business cycle and monetary policy analysis in a structural sticky-price model of the euro area" by M. Casares, March 2001.
- 50 "Employment and productivity growth in service and manufacturing sectors in France, Germany and the US" by T. von Wachter, March 2001.
- 51 "The functional form of the demand for euro area MI" by L. Stracca, March 2001.
- 52 "Are the effects of monetary policy in the euro area greater in recessions than in booms?" by G. Peersman and F. Smets.
- 53 "An evaluation of some measures of core inflation for the euro area" by J.-L. Vega and M.A. Wynne, April 2001.
- 54 "Assessment criteria for output gap estimates" by G. Camba-Méndez and D. R. Palenzuela, April 2001.
- 55 "Modelling the demand for loans to the private sector in the euro area" by A. Calza, G. Gartner and J. Sousa, April 2001.

56 "Stabilization policy in a two country model and the role of financial frictions" by E. Faia, April 2001.